# REGULARITY CONDITIONS IN THE REALISABILITY PROBLEM WITH APPLICATIONS TO POINT PROCESSES AND RANDOM CLOSED SETS

## BY RAPHAEL LACHIEZE-REY AND ILYA MOLCHANOV<sup>1</sup>

#### Université Paris Descartes and University of Bern

We study existence of random elements with partially specified distributions. The technique relies on the existence of a positive extension for linear functionals accompanied by additional conditions that ensure the regularity of the extension needed for interpreting it as a probability measure. It is shown in which case the extension can be chosen to possess some invariance properties.

The results are applied to the existence of point processes with given correlation measure and random closed sets with given two-point covering function or contact distribution function. It is shown that the regularity condition can be efficiently checked in many cases in order to ensure that the obtained point processes are indeed locally finite and random sets have closed realisations.

**1. Introduction.** Defining the distribution of a random element  $\xi$  in a topological space  $\mathcal{X}$  is equivalent to specialising the expected values for all bounded continuous functions  $g(\xi)$ . These expected values define a linear functional  $\Phi(g) = Eg(\xi)$  on the space of bounded continuous functions  $g: \mathcal{X} \mapsto \mathbb{R}$ . It is well known that a functional  $\Phi$  indeed corresponds to a random element if and only if  $\Phi$  is positive [i.e.,  $\Phi(g) \ge 0$  if g is nonnegative] and upper semi-continuous [i.e.,  $\Phi(g_n) \downarrow 0$  if  $g_n \downarrow 0$ ]; see, for example, [36].

Below we consider the case of functional  $\Phi$  defined only on some functions on  $\mathcal{X}$  and address the *realisability* of  $\Phi$ , that is, the mere existence of a random element  $\xi$  such that  $\Phi(g) = Eg(\xi)$  for g from the chosen family G of functions. The uniqueness is not on the agenda, since typically the family G will not suffice to uniquely specify the distribution of  $\xi$ . A classical example of this setting is the existence of a probability distribution with given marginals; see [9]. The present paper focuses on some geometric instances of the problem. We will see that in most cases the answer to the existence problem consists of the two main steps.

Received March 2011; revised September 2013.

<sup>&</sup>lt;sup>1</sup>Supported by Swiss National Foundation Grant No. 200021-126503. The revision was done while the second author held the Chair of Excellence at the University Carlos III of Madrid supported by the Santander Bank.

*MSC2010 subject classifications.* Primary 60D05; secondary 28C05, 46A40, 47B65, 60G55, 74A40, 82D30.

*Key words and phrases.* Point process, correlation measure, random closed set, two-point covering probability, contact distribution function, realisability.

1. (*Positivity*) Checking the positivity condition on  $\Phi$ —in most cases this requires checking a system of inequalities, which is a serious (but unavoidable) computational burden.

2. (*Regularity*) Ensuring that the extended functional is regular (namely, upper semi-continuous) and so defines a  $\sigma$ -additive measure.

The first step ensures that it is possible to extend functional  $\Phi$  positively from a certain family of functions to a wider family. In this work, we put the emphasis on the latter step—checking the regularity condition, leaving aside the computational difficulties arising from validating the positivity assumption.

The use of positive extension techniques (that goes back to Kantorovitch) in the framework of stochastic geometry was pioneered by Kuna, Lebowitz and Speer [13] in application to point processes, which greatly inspired the current work. In this paper, we establish the general nature of an idea proposed in [13] and show how it leads to various further realisability results. The new idea is to introduce an additional function, what we call the regularity modulus, and to formulate sufficient and necessary conditions in terms of a positive extension of a functional onto the linear space containing the regularity modulus and requiring only a priori integrability of the regularity modulus.

We concentrate on two basic examples of the realisability problem: the existence of point processes with given correlation (factorial moment) measure and the existence of a random closed set with given two-point coverage probabilities or contact distribution functions. The introduction to the realisability issue for point processes is available in several papers by Kuna, Lebowitz and Speer [12, 13]; see also Section 3 of this paper. The realisability problem for random closed sets has been widely studied in physics and material science literature; see [8, 17, 31, 33, 34] and in particular the comprehensive monograph by Torquato [32] and a recent survey by Quintanilla [24]. If  $\xi$  is a random closed set (see Section 4 for formal definitions) in a locally compact metric space X, its *one-point covering functions* is defined by

$$p_x = \mathbf{P}\{x \in \xi\}, \qquad x \in \mathbb{X}$$

It is easy to characterise all one-point covering functions of random closed sets as follows.

THEOREM 1.1. A function  $p_x$ ,  $x \in \mathbb{X}$ , with values in [0, 1] is the one-point covering function of a random closed set if and only if p is upper semi-continuous.

The upper semi-continuity of the one-point covering function of a random closed set  $\xi$  is a straightforward consequence of the upper semi-continuity property of the capacity functional of a random closed set; see [22], Section 1.1.2. Conversely, the function p from the theorem is realised (e.g.) as the one-point

covering function of the random set  $\xi = \{x : p_x \ge v\}$  where v is a uniformly distributed variable (the details are left to the reader).

It is considerably more complicated to characterise two-point covering functions

$$p_{x,y} = \mathbf{P}\{x, y \in \xi\}, \qquad x, y \in \mathbb{X}.$$

In view of applications to modelling of random media, it is often assumed that  $\xi$  is a stationary set in  $\mathbb{R}^d$ , so that the one-point covering function is constant and the two-point covering function  $p_{x,y}$  depends only on x - y. Since a random closed set can be considered as an upper semi-continuous indicator function, the realisability problem for the two-point covering function can be rephrased as follows:

Characterise covariance functions of (stationary) upper semi-continuous random functions with values in  $\{0, 1\}$ .

These covariances are obviously a sub-family of positive semi-definite functions. Without the upper semi-continuity requirement, this problem, of combinatorial nature, was solved by McMillan [21] and Shepp [27, 28] using the extension argument from [9]. More exactly, they normalised indicators by letting them take values +1 or -1 and assumed that the mean is zero. Their result does not rely on the topological structure of the underlying space and so does not necessarily lead to an upper semi-continuous indicator function.

EXAMPLE 1.2. Let  $p_{x,y} = \frac{1}{4}$  and let  $p_x = \frac{1}{2}$  for all  $x, y \in \mathbb{R}$ . While this twopoint covering function corresponds, for example, to the indicator field with independent values, it cannot be obtained as the two-point covering function of a random closed set; see Proposition 4.4.

Even leaving aside the upper semi-continuity property, the McMillan–Shepp condition involves a family of corner-positive matrices, which is poorly understood. As a result, its practical use to check the realisability for random media is rather limited. A number of authors have attempted to come up with simpler (but only necessary) conditions; see, for example, [8, 19, 24, 33]. Another set of conditions for joint distributions of binary random variables is formulated in [26] in terms of the corresponding copulas.

The realisability problem can be also posed for point processes in terms of their moment measures. In case of moment measures of arbitrary order, it has been solved by Lenard [15, 16]. The case of moment measures up to the second order has been studied by Kuna, Lebowitz and Speer [12], whose recent paper [13] contains (among other results) a *complete* solution of this realisability problem for point processes with finite third-order moments and hard-core type conditions with fixed exclusion distance. The results of [13] can be extended to higher order moment measures, as was explicitly indicated there. Again, the positivity condition of [13] is extremely difficult to verify, even more complicated than the original

condition for point processes because of new polynomial functionals involved in the positivity condition.

The paper is organised as follows. Section 2 presents a series of general results on regular extensions and also invariant extensions (relevant for the existence of stationary random elements). These results form the theoretical backbone of our study, and are new even in the abstract setting of extending general positive linear functionals.

Section 3 presents a number of realisability conditions for correlation measures of point processes that considerably extend the results of [13] by relaxing the moment and hardcore conditions. One of our most important results is Theorem 3.3 that shows how to split the positivity and regularity conditions, so that the latter can be efficiently checked. The importance of the packing number in relation to realisability conditions for hard-core point processes is also explained.

Section 4 deals with the realisability problem for two-point covering probabilities of random sets. The closedness of the corresponding random set can be ensured by imposing appropriate regularity conditions. Section 5 addresses a further variant of the realisability problem that involves contact distribution functions of random sets.

The notational convention is that the carrier space is denoted as  $\mathbb{X}$  (e.g.,  $\mathbb{R}$ ,  $\mathbb{R}^d$ ), points in the carrier space are x, y, subsets of carrier spaces are denoted by capitals X, Y, F (while Y is reserved for counting measures identified with corresponding support sets), the families of sets (or families of counting measures) as  $\mathcal{X}, \mathcal{N}, \mathcal{F}$ (while in Section 2  $\mathcal{X}$  denotes also a rather general space) and random element in these spaces (random sets or point processes) as  $\xi$ , real functions acting on  $\mathcal{X}, \mathcal{N}, \mathcal{F}$  are g, v and families of such functions are G, E, V, a functional on G, E, V is denoted by  $\Phi$ , real numbers are denoted by  $t, r, \lambda$ , while c denotes a generic constant and at the same time the corresponding constant function.

**2. Extending positive functionals.** Fundamental results about the extension of positive operators form the heart of our main results, and are necessary to understand the machinery of the proofs. Nevertheless, the results of the subsequent sections can be understood without Section 2, with the exception of Definition 2.5.

2.1. General extension theorems. Consider a vector lattice E, that is a linear space with a partial order and such that for any  $v_1, v_2 \in E$  their maximum  $v_1 \lor v_2$  also belongs to E. The absolute value |v| of v is defined as the sum of  $v \lor 0$  and  $(-v) \lor 0$ .

Let G be a *vector subspace* of E, which is not necessarily a lattice itself, that is G may be not closed with respect to the maximum operation. We say that G *majorises* E if each  $v \in E$  satisfies  $|v| \leq g$  for some  $g \in G$ . A real-valued functional  $\Phi$  defined on E (resp., G) is said to be *positive* if  $\Phi(v) \geq 0$  whenever  $v \geq 0$  and  $v \in E$  (resp.,  $v \in G$ ). A functional defined on E is said to be an *extension* of  $\Phi: G \mapsto \mathbb{R}$  if it coincides with  $\Phi$  on G. The extended  $\Phi$  is always denoted by the same letter. The following result about extension of positive functionals goes back to Kantorovich.

THEOREM 2.1 (See [1], Theorem 8.12 and [35], Theorem X.3.1). Assume that G is a majorising vector subspace of a vector lattice E. Then each positive linear functional on G admits a positive extension on the whole E.

If G is a lattice itself, then it is possible to gain much more control over the extension of  $\Phi$ , for example, a continuous functional admits a continuous extension, see [35], Section X.5. On the contrary, very little is known about regularity properties of the extension if G is not a lattice.

In the following, we assume that G and E are families of functions g on a certain space  $\mathcal{X}$ . If G contains constant functions, the positivity of  $\Phi$  over G can be equivalently formulated as

(2.1) 
$$\Phi(\mathbf{g}) \ge \inf_{X \in \mathcal{X}} \mathbf{g}(X).$$

This equivalence is a particular case of the following result for  $\chi = 0$  [replace g with -g in (2.2)].

PROPOSITION 2.2. Assume that vector space G contains constant functions and denote by  $G \setminus \mathbb{R}$  the family of nonconstant functions from G. If  $\chi$  is any nonnegative function on  $\mathcal{X}$ , then a linear functional  $\Phi$  on G admits a positive extension on  $G + \mathbb{R}\chi$  with  $\Phi(\chi) = r$  if and only if

(2.2) 
$$r = \sup_{\mathsf{g}\in\mathsf{G},\mathsf{g}\leq\chi} \Phi(\mathsf{g}) = \sup_{\mathsf{g}\in\mathsf{G}\setminus\mathbb{R}} \inf_{X\in\mathcal{X}} [\chi(X) - \mathsf{g}(X)] + \Phi(\mathsf{g}) < \infty.$$

PROOF. Since every element of G can be written c + g with  $g \in G \setminus \mathbb{R}$  and  $c \in \mathbb{R}$ , the left-hand side of (2.2) equals

$$r = \sup_{\mathsf{g}\in\mathsf{G}\backslash\mathbb{R}} \sup_{c\in\mathbb{R}: c+\mathsf{g}\leq\chi} c+\Phi(\mathsf{g}) = \sup_{\mathsf{g}\in\mathsf{G}} c_{\mathsf{g}} + \Phi(\mathsf{g}),$$

where  $c_g = \inf_{X \in \mathcal{X}} (\chi - g)(X)$  is the largest *c* such that  $c + g \le \chi$ , which yields the equality in (2.2).

The necessity of (2.2) is straightforward because  $r \leq \Phi(\chi) < \infty$ . For the sufficiency, assume that (2.2) holds. The proof consists in checking that assigning the value  $\Phi(\chi) = r$  yields a positive extension on  $G + \mathbb{R}\chi$ . Let us first prove that  $\Phi$  is positive on G. If some  $g \leq 0$  satisfies  $\Phi(g) > 0$ , then  $\Phi(tg) \uparrow \infty$  as  $t \to \infty$  whereas  $tg \leq \chi$ , which contradicts (2.2).

Let  $g + \lambda \chi \ge 0$  for  $\lambda \ne 0$  and  $g \in G$ . If  $\lambda > 0$ , then  $-\lambda^{-1}g \le \chi$ , whence  $\Phi(-\lambda^{-1}g) \le r$  and  $\Phi(g+\lambda\chi) \ge -\lambda r + \lambda \Phi(\chi) = 0$ . If  $\lambda < 0, -\lambda^{-1}g \ge \chi$  whence  $-\lambda^{-1}g$  is larger than any  $g' \le \chi$ , and

$$\Phi(-\lambda^{-1}g) \ge \sup_{g' \in G, g' \le \chi} \Phi(g') = r$$

by monotonicity of  $\Phi$  on G. Hence,  $\Phi(g + \lambda \chi) \ge -\lambda r + \lambda \Phi(\chi) = 0$ .  $\Box$ 

The advantage of the latter condition in (2.2) consists in the explicit reference to the space  $\mathcal{X}$  where random elements lie instead of checking the inequality  $g \leq \chi$ .

2.2. Regularity conditions and distributions of random elements. Let E be a certain family of functions  $v: \mathcal{X} \mapsto \mathbb{R}$  defined on a space  $\mathcal{X}$  with lattice operation being the pointwise maximum and the corresponding partial order.

THEOREM 2.3 (Daniell, see [4], Section 4.5 and [11], Theorem 14.1). Let a vector lattice  $\mathsf{E}$  consist of real-valued functions on  $\mathcal{X}$  and let  $\mathsf{E}$  contain constants. If  $\Phi$  is a positive functional on  $\mathsf{E}$  such that  $\Phi(\mathsf{v}_n) \downarrow 0$  for each sequence  $\mathsf{v}_n \downarrow 0$  and  $\Phi(1) = 1$ , then there exists a unique random element  $\xi$  in  $\mathcal{X}$ , measurable with respect to the  $\sigma$ -algebra generated by all functions from  $\mathsf{E}$ , such that  $\Phi(\mathsf{v}) = \mathbf{E}\mathsf{v}(\xi)$  for all  $\mathsf{v} \in \mathsf{E}$ .

In view of the positivity of  $\Phi$ , the condition imposed on  $\Phi$  is equivalent to its upper semi-continuity on E. In this paper, we start with a functional  $\Phi$  defined on a vector sub-space  $\mathbf{G} \subset \mathbf{E}$  and discuss the existence of a random element  $\xi \in \mathcal{X}$  such that  $\Phi(\mathbf{g}) = \mathbf{E}\mathbf{g}(\xi)$  for all  $\mathbf{g} \in \mathbf{G}$ . In this case,  $\Phi$  is said to be *realisable* as a probability distribution on  $\mathcal{X}$ .

ASSUMPTION 2.4. The vector space G of functions on  $\mathcal{X}$  contains constants and, for each  $g_1, g_2 \in G$ , there exists a  $g \in G$  such that  $(g_1 \vee g_2) \leq g$ .

From now on assume that  $\mathcal{X}$  is a completely regular topological space, that is, each closed set and each singleton disjoint from it can be separated by a continuous function.

DEFINITION 2.5. Given a vector space G of functions on  $\mathcal{X}$ , a *regularity modulus* on  $\mathcal{X}$  is a lower semi-continuous function  $\chi : \mathcal{X} \mapsto [0, \infty]$  such that

(2.3) 
$$\mathcal{H}_{g} = \left\{ X \in \mathcal{X} : \chi(X) \le g(X) \right\}$$

is relatively compact for each  $g \in G$  (if all  $g \in G$  are bounded,  $\chi$  is a regularity modulus if and only if it has compact level sets).

Examples of regularity moduli are given in Sections 3 and 4. A measurable function  $v: \mathcal{X} \mapsto \mathbb{R}$  is said to be  $\chi$ -regular if v is continuous on  $\mathcal{H}_g$  for each g in G. Each continuous function is trivially  $\chi$ -regular. The proof of the following central result is based on the ideas from the proof of [13], Theorem 3.14. It should be noted that our result entails not only the realisability, but also provides a bound for the expected value of the regularity modulus. It also holds on not necessarily completely regular space  $\mathcal{X}$  if the regularity modulus is continuous or otherwise without the explicit bound on  $\mathbf{E}_{\chi}(\xi)$ .

THEOREM 2.6. Consider a vector space G of functions on  $\mathcal{X}$  satisfying Assumption 2.4 and such that each g from G is  $\chi$ -regular for a regularity modulus  $\chi$ .

Let  $\Phi$  be a linear functional on G with  $\Phi(1) = 1$ . Then, for any given  $r \ge 0$ , there exists a Borel random element  $\xi$  in  $\mathcal{X}$  such that

(2.4) 
$$\begin{cases} \mathbf{Eg}(\xi) = \Phi(\mathbf{g}) & \text{for all } \mathbf{g} \in \mathbf{G}, \\ \mathbf{E}\chi(\xi) \le r, \end{cases}$$

*if and only if* 

(2.5) 
$$\sup_{\mathbf{g}\in\mathbf{G},\mathbf{g}\leq\chi}\Phi(\mathbf{g})\leq r.$$

PROOF. Condition (2.5) is necessary because  $g \le \chi$  implies  $\Phi(g) = Eg(\xi) \le E\chi(\xi) \le r$ .

Sufficiency. Let E be the family of all  $\chi$ -regular functions v that satisfy  $v \leq g$  for some  $g \in G$ . Each function  $v \in E$  is Borel measurable. Note that E contains all bounded continuous functions that generate the Baire  $\sigma$ -algebra on  $\mathcal{X}$  being in general a sub- $\sigma$ -algebra of the Borel one. For each  $v_1, v_2 \in E$ , the function  $v_1 \lor v_2$  is  $\chi$ -regular and is majorised by  $g_1 \lor g_2$ , where  $g_1, g_2 \in G$  majorise  $v_1$  and  $v_2$ , respectively. In view of Assumption 2.4, E is a lattice.

Without loss of generality, assume that the supremum in (2.5) equals *r*. By Proposition 2.2,  $\Phi$  is positive on G and can be positively extended onto  $\mathbf{G} + \mathbb{R}\chi$ with  $\Phi(\chi) = r$ , and further on to  $\mathbf{E} + \mathbb{R}\chi$  by Theorem 2.1. It remains to prove that the obtained extension satisfies conditions of Theorem 2.3. For that, we use an argument similar to that of [13]. First, restrict the obtained functional  $\Phi$  onto E. Assume that  $\chi$  is strictly positive. Consider a sequence  $\{\mathbf{v}_n, n \ge 1\} \subset \mathbf{E}$  such that  $\mathbf{v}_n \downarrow 0$ . For each *n*, let  $g_n$  be a function of G such that  $\mathbf{v}_n \le g_n$ . Take  $\varepsilon > 0$ . Then  $\mathcal{K}_n = \{X : \mathbf{v}_n(X) \ge \varepsilon \chi(X)\}$  is a subset of relatively compact  $\mathcal{H}_{g_n/\varepsilon}$ , since  $\chi$  is a regularity modulus. Since  $\mathbf{v}_n$  is continuous on  $\mathcal{H}_{g_n/\varepsilon}$ , the set  $\mathcal{K}_n$  is closed and, therefore, compact. The pointwise convergence  $\mathbf{v}_n \downarrow 0$  yields that  $\bigcap_n \mathcal{K}_n = \emptyset$  (recall that  $\chi$  is strictly positive). Since  $\{\mathcal{K}_n\}$  is a decreasing sequence of compact sets,  $\mathcal{K}_{n_0} = \emptyset$  for some  $n_0$ , whence  $\mathbf{v}_n(X) < \varepsilon \chi(X)$  for sufficiently large *n*. The positivity of  $\Phi$  on  $\mathbf{E} + \mathbb{R}\chi$  implies  $\Phi(\mathbf{v}_n) \le \varepsilon \Phi(\chi) = \varepsilon r$ , whence  $\Phi(\mathbf{v}_n) \downarrow 0$ . Theorem 2.3 yields the existence of a random element  $\xi$  in  $\mathcal{X}$  such that  $\Phi(\mathbf{v}) = \mathbf{E}\mathbf{v}(\xi)$ for all  $\mathbf{v} \in \mathbf{E}$ .

Since  $\chi$  is lower semi-continuous and  $\mathcal{X}$  is completely regular, it can be pointwisely approximated from below by a sequence  $\{v_n\}$  of nonnegative continuous functions; see [2], Chapter 9. Then  $\tilde{v}_n = \min(n, v_n)$  belongs to E and also approximates  $\chi$  from below, so that  $\mathbf{E}\tilde{v}_n(\xi) = \Phi(\tilde{v}_n) \leq \Phi(\chi) = r$ , while the monotone convergence theorem yields

$$\mathbf{E}\chi(\xi) = \lim_{n \to \infty} \mathbf{E}\tilde{\mathsf{v}}_n(\xi) \le r.$$

If  $\chi$  is not strictly positive, it suffices to apply the above argument to  $\chi' = 1 + \chi$  and use the linearity of  $\Phi$ .  $\Box$ 

Condition (2.5), equivalent to (2.2), is expressed solely in terms of the values taken by  $\Phi$  on G and, therefore, yields a self-contained solution of the realisability problem. It is not easy to check in general, but if  $\chi$  can be approximated by functions  $\chi_n \in G$ ,  $n \ge 1$ , then it is possible to "split" (2.5) into the positivity condition on  $\Phi$  and the uniform boundedness of  $\Phi(\chi_n)$ ,  $n \ge 1$ . This idea is used successfully in several different frameworks, which justify the abstract setting of Theorem 2.6: in Section 3 for point processes (see Theorem 3.1), in Section 4.4 for random closed sets (see Theorem 4.9) and in [6] in the framework of random measurable sets with the regularity modulus being the perimeter of a set.

The realisability problem is particularly simple if  $\mathcal{X}$  is compact and G consists of continuous functions. Then, for identically vanishing  $\chi$ , Theorem 2.6 yields the following result, which is similar to the Riesz–Markov theorem; see [11].

COROLLARY 2.7. Let  $\mathcal{X}$  be a compact space with its Borel  $\sigma$ -algebra. Consider a vector space G containing constants such that each  $g \in G$  is continuous and a map  $\Phi: G \mapsto \mathbb{R}$  such that  $\Phi(1) = 1$ . Then there exists a random element  $\xi$  in  $\mathcal{X}$  such that  $Eg(\xi) = \Phi(g)$  for all  $g \in G$  if and only if  $\Phi$  is a linear positive functional on G.

It should be noted that the complete regularity assumption on  $\mathcal{X}$  is not needed if the regularity modulus  $\chi$  is continuous.

2.3. *Passing to the limit*. The following result shows that the family of all random elements that realise  $\Phi$  in the sense of (2.4) is weakly compact.

THEOREM 2.8. Assume that G satisfies Assumption 2.4 and consists of continuous functions on a Polish space  $\mathcal{X}$  with regularity modulus  $\chi$ . Let  $\Phi$  be a linear positive functional on G. Then the family  $\mathfrak{M}$  of all Borel random elements  $\xi$  that satisfy (2.4) for any given  $r \geq 0$  is compact in the weak topology.

**PROOF.** Since  $\chi$  is a regularity modulus, the set  $\mathcal{H}_{r/\varepsilon}$  is compact. By Markov's inequality,

$$\mathbf{P}\{\xi \notin \mathcal{H}_{r/\varepsilon}\} = \mathbf{P}\{\chi(\xi) > r/\varepsilon\} \leq \varepsilon,$$

for all  $\xi \in \mathfrak{M}$ , so that  $\mathfrak{M}$  is tight.

Let  $\{\xi_n, n \ge 1\}$  be random elements from  $\mathfrak{M}$ . Assume that  $\xi_n$  converges weakly to some  $\xi$ . Without loss of generality, assume that the  $\xi_n$ 's are defined on the same probability space and converge almost surely to  $\xi$ . Since  $\chi$  is nonnegative, Fatou's lemma yields

$$r \geq \liminf \mathbf{E}\chi(\xi_n) \geq \mathbf{E}\liminf \chi(\xi_n) \geq \mathbf{E}\chi(\limsup \xi_n) = \mathbf{E}\chi(\xi),$$

where the lower semi-continuity of  $\chi$  also has been used.

Take an arbitrary  $g \in G$  and define  $\mathcal{H}_{\lambda g}$  as in (2.3). Let  $g^+(X) = \max(g(X), 0)$  be the positive part of g. Then, for  $\lambda > 0$ ,

$$\mathbf{Eg}^{+}(\xi_{n}) = \mathbf{Eg}^{+}(\xi_{n})\mathbb{1}_{\xi_{n}\notin\mathcal{H}_{\lambdag}} + \mathbf{Eg}^{+}(\xi_{n})\mathbb{1}_{\xi_{n}\in\mathcal{H}_{\lambdag}}.$$

Since g is continuous,  $\mathcal{H}_{\lambda g}$  is closed (and compact), so that if  $\xi_n \in \mathcal{H}_{\lambda g}$  for infinitely many *n*, then also  $\xi \in \mathcal{H}_{\lambda g}$ . Furthermore,  $\lambda g$  and also g itself, are continuous and bounded on  $\mathcal{H}_{\lambda g}$ , so that Fatou's lemma yields

$$\begin{split} \limsup \mathbf{E} \mathbf{g}^+(\xi_n) \mathbb{1}_{\xi_n \in \mathcal{H}_{\lambda g}} &\leq \mathbf{E} \limsup \left( \mathbf{g}^+(\xi_n) \mathbb{1}_{\xi_n \in \mathcal{H}_{\lambda g}} \right) \\ &\leq \mathbf{E} \mathbf{g}^+(\xi) \mathbb{1}_{\xi \in \mathcal{H}_{\lambda g}} \leq \mathbf{E} \mathbf{g}^+(\xi). \end{split}$$

Thus,

$$\limsup \operatorname{Eg}^+(\xi_n) \leq \operatorname{E}\frac{\chi(\xi_n)}{\lambda} + \operatorname{Eg}^+(\xi) \leq \frac{r}{\lambda} + \operatorname{Eg}^+(\xi).$$

Since  $\lambda$  is arbitrary,

$$\limsup \operatorname{Eg}^+(\xi_n) \leq \operatorname{Eg}^+(\xi).$$

Since  $g^+$  is nonnegative, Fatou's lemma yields that  $Eg^+(\xi_n) \to Eg^+(\xi)$ . By applying the same argument to the function (-g),  $\lim Eg(\xi_n) = Eg(\xi)$ , so that  $Eg(\xi) = \Phi(g)$  for all  $g \in G$ . Therefore,  $\xi \in \mathfrak{M}$ .  $\Box$ 

The following result concerns realisability of pointwise limits of linear functionals. Special conditions of this type for correlation measures of point processes are given in [13], Section 3.4.

THEOREM 2.9. Let  $\{\Phi_n, n \ge 1\}$  be a sequence of linear positive functionals on a space G that satisfies the assumptions of Theorem 2.8. Assume that

(2.6) 
$$\liminf_{n} \sup_{g \in G, g \leq \chi} \Phi_n(g) < \infty.$$

If  $\Phi_n(g) \to \Phi(g)$  for all  $g \in G$ , then  $\Phi$  is realisable as a random element  $\xi$  satisfying (2.4) and such that  $\xi$  is the weak limit of random elements realising  $\Phi_{n_k}$  for a subsequence  $n_k$ .

PROOF. By passing to a subsequence, it suffices to assume that (2.6) holds for the limit instead of the lower limit. Let  $\xi_n$  be a random element that realises  $\Phi_n$ . If *r* is larger than the limit of (2.6), then  $\mathbf{P}\{\xi_n \notin \mathcal{H}_{r/\varepsilon}\} \leq \varepsilon$ , so that  $\{\xi_n\}$  is a tight sequence. Without loss of generality, assume that  $\xi_n$  weakly converges to a random element  $\xi$ .

The pointwise convergence of  $\Phi_n$  yields that  $\mathbf{Eg}(\xi_n) \to \Phi(g)$  for all  $g \in G$ . Now the arguments from the proof of Theorem 2.8 can be used to show that  $\mathbf{Eg}(\xi_n) \to \mathbf{Eg}(\xi)$ , so that  $\mathbf{Eg}(\xi) = \Phi(g)$  for all  $g \in G$ , that is,  $\xi$  indeed satisfies (2.4).  $\Box$ 

2.4. *Invariant extension*. Consider an Abelian group  $\Theta$  of continuous transformations acting on  $\mathcal{X}$ . For a function v on  $\mathcal{X}$ , define

$$(\theta \mathsf{v})(X) = \mathsf{v}(\theta X), \qquad \theta \in \Theta, X \in \mathcal{X}.$$

A functional  $\Phi$  is said to be  $\Theta$ -invariant if, for each  $\theta \in \Theta$  and v from the domain of definition of  $\Phi$ ,  $\Phi(\theta v)$  is defined and equal to  $\Phi(v)$ .

A Borel random element  $\xi$  in  $\mathcal{X}$  is said to be  $\Theta$ -stationary if, for each  $\theta \in \Theta$ ,  $\theta \xi$  has the same distribution as  $\xi$ . A variant of the following result for correlation measures of point processes is given in [13], Theorem 4.3.

THEOREM 2.10. Assume that G is a  $\Theta$ -invariant space satisfying Assumption 2.4 and consisting of  $\chi$ -regular functions. Furthermore, assume that at least one of the following conditions holds:

(i) G consists of continuous functions and  $\chi$  is pointwisely approximated from below by a monotone sequence of functions  $g_n \in G$ ,  $n \ge 1$ .

(ii)  $\chi$  is  $\Theta$ -invariant.

Let  $\Phi$  be a  $\Theta$ -invariant functional on G. Then, for every given  $r \ge 0$ , there exists a  $\Theta$ -stationary random element  $\xi$  in  $\mathcal{X}$  satisfying (2.4) if and only if (2.5) holds.

PROOF. (i) As in [13], Proposition 4.1, the proof consists in checking hypotheses of the Markov–Kakutani fixed-point theorem. Let  $\mathfrak{M}$  be the family of random elements  $\xi$  that realise  $\Phi$  on G, and satisfy  $\mathbf{E}\chi(\theta\xi) \leq r$  for every  $\theta \in \Theta$ . The family  $\mathfrak{M}$  is easily seen to be convex with respect to addition of measures, it is compact by Theorem 2.8, and  $\Theta$ -invariant, since  $\Phi$  is  $\Theta$ -invariant on G. It remains to prove that  $\mathfrak{M}$  is not empty.

In view of (2.5), it is possible to extend  $\Phi$  positively onto  $\mathbf{G} + \mathbb{R}\chi$ , so that  $\mathbf{E}\chi(\xi) \leq r$ . The  $\Theta$ -invariance of  $\Phi$  on  $\mathbf{G}$  together with the monotone convergence theorem imply that  $\mathbf{E}\chi(\theta\xi) = \mathbf{E}\chi(\xi) \leq r$ , whence  $\xi \in \mathfrak{M}$ .

(ii) By Proposition 2.2, we can extend  $\Phi$  positively onto the  $\Theta$ -invariant vector space  $V = G + \mathbb{R}\chi$ . Since  $\Phi$  is  $\Theta$ -invariant on G, we have  $\Phi(\theta(g + t\chi)) = \Phi(\theta g) + t\Phi(\theta \chi) = \Phi(g + t\chi)$  for  $g + t\chi$  in V, whence  $\Phi$  is  $\Theta$ -invariant on V. According to [29], Theorem 3,  $\Phi$  admits a positive  $\Theta$ -invariant extension to the space  $E + \mathbb{R}\chi$ , defined like in the proof of Theorem 2.6. The restriction of the obtained functional onto E corresponds to a random element  $\xi$  in  $\mathcal{X}$  that verifies (2.4) and satisfies  $E(\theta v)(\xi) = \Phi(\theta v) = \Phi(v) = Ev(\xi), \theta \in \Theta$ , for v in E. Since E contains all bounded continuous functions on  $\mathcal{X}, \theta \xi$  and  $\xi$  are identically distributed for all  $\theta \in \Theta$ .  $\Box$ 

#### 3. Correlation measures of point processes.

3.1. *Framework and main results*. Let  $\mathcal{N}$  be the family of locally finite counting measures on a locally compact complete separable metric space  $\mathbb{X}$ . We denote the support of  $Y \in \mathcal{N}$  by the same letter Y, so that  $x \in Y$  means  $Y(\{x\}) \ge 1$ .

Equip  $\mathcal{N}$  with the vague topology, see [3], Chapter 7, so that  $\mathcal{N}$  is metric and so completely regular. A random element  $\xi$  in  $\mathcal{N}$  with the corresponding Borel  $\sigma$ -algebra is called a *point process*. Denote by  $\mathcal{N}_0$  the family of *simple* counting measures, that is, those which do not attach mass 2 or more to any given point. If  $\xi$ is simple, that is,  $\xi \in \mathcal{N}_0$  a.s., then  $\xi$  can be identified with a locally finite random set in  $\mathbb{X}$ , which is also denoted by  $\xi$ .

For a real function *h* on  $\mathbb{X} \times \mathbb{X}$  and counting measure  $Y = \sum_i \delta_{x_i}$  given by the sum of Dirac measures, define

$$\mathsf{g}_h(Y) = \sum_{x_i, x_j \in Y, i \neq j} h(x_i, x_j),$$

whenever the series absolutely converges, the empty sum being 0. Note that the sum in the right-hand side is taken over all pairs of distinct points from the support of Y, where multiple points appear several times according to their multiplicities. The value  $g_h(Y)$  is necessarily finite if h is bounded and has a bounded support. The value  $g_h(Y)$  is termed in [13] the quadratic polynomial of Y, while polynomials of order  $n \ge 1$  are sums of functions of n points of the process, and are constants if n = 0.

Let G be the vector space formed by constants and functions  $g_h$  for *h* from the space  $\mathscr{C}_0$  of symmetric continuous functions with compact support. Note that G satisfies Assumption 2.4, since

$$(c_1 + \mathsf{g}_{h_1}) \lor (c_2 + \mathsf{g}_{h_2}) \le c_1 \lor c_2 + \mathsf{g}_{h_1 \lor h_2} \in \mathsf{G}$$

for all  $c_1, c_2 \in \mathbb{R}$  and  $h_1, h_2 \in \mathcal{C}_0$ . Furthermore, each  $g_h$  is continuous in the vague topology, and so is  $\chi$ -regular for any regularity modulus  $\chi$ .

Assume that  $\xi$  has locally finite second moment, that is,  $\mathbf{E}\xi(A)^2$  is finite for each bounded A. The *correlation measure*  $\rho$  (also called the second factorial moment measure) of a point process  $\xi$  is a measure on  $\mathbb{X} \times \mathbb{X}$  that satisfies

(3.1) 
$$\int_{\mathbb{X}\times\mathbb{X}} h(x, y)\rho(dx\,dy) = \mathbf{E}\mathbf{g}_h(\xi)$$

for each  $h \in \mathscr{C}_{0}$ ; see [3], Section 5.4 and [30], Section 4.3. The left-hand side defines a linear functional  $\Phi(g_{h})$  on  $g_{h} \in G$ .

Let  $\mathcal{X}$  be a subset of  $\mathcal{N}$ , which may be  $\mathcal{N}$  itself. Recall that a subset of a completely regular space is completely regular, see [14], Theorem 14.I.2. Given a symmetric locally finite measure  $\rho$  on  $\mathbb{X} \times \mathbb{X}$ , the *realisability problem* amounts to the existence of a point process  $\xi$  with realisations from  $\mathcal{X}$  and with correlation measure  $\rho$ , so that  $\Phi(\mathbf{g}_h) = \mathbf{E}\mathbf{g}_h(\xi)$  for all  $h \in \mathcal{C}_0$ .

By (2.1), the positivity of  $\Phi$  means

(3.2) 
$$\Phi(\mathbf{g}_h) \ge \inf_{Y \in \mathcal{X}} \mathbf{g}_h(Y)$$

for all  $h \in \mathcal{C}_0$ . Then it is clear that the positivity of  $\Phi$  is necessary for its realisability. If  $\mathcal{X}$  is compact in the vague topology, then Corollary 2.7 applies and the positivity condition (3.2) is necessary and sufficient for the realisability of  $\rho$ .

However, in general the positivity condition alone is not sufficient for the realisability; see [12], Example 3.12. In the following, we find another condition that is not directly related to the positivity, but together with the positivity, is necessary and sufficient for the realisability.

As an introduction, let us present our results for X being a subset of the Euclidean space  $\mathbb{R}^d$ . For  $\varepsilon \ge 0$ , define

$$\chi_{\varepsilon}(Y) = \sum_{x, y \in Y, x \neq y} \|x - y\|^{-d - \varepsilon}, \qquad Y \in \mathcal{N},$$

which is later acknowledged as being a regularity modulus (see Definition 2.5) if  $\varepsilon \neq 0$ . Note that  $\chi_{\varepsilon}(Y)$  is infinite if Y has multiple points. The tools developed in this paper enable us to resolve the original realisability problem with a supplementary regularity condition involving  $\chi_{\varepsilon}$ .

THEOREM 3.1. (i) Let X be a compact subset of  $\mathbb{R}^d$  without isolated points. A symmetric finite measure  $\rho(dx dy)$  on  $X \times X$  is the correlation measure of a simple point process  $\xi \subset X$  such that  $\mathbf{E}\chi_0(\xi) < \infty$  if and only if  $\Phi$  given by the left-hand side of (3.1) is positive and

$$\int_{\mathbb{X}\times\mathbb{X}} \|x-y\|^{-d} \rho(dx\,dy) < \infty.$$

(ii) Let  $\rho$  be a symmetric locally finite measure on  $\mathbb{R}^d \times \mathbb{R}^d$  such that  $\rho((A + x) \times (B + x)) = \rho(A \times B)$  for all  $x \in \mathbb{R}^d$  and measurable sets A and B. Then there exists a simple stationary point process  $\xi$  with correlation measure  $\rho$ , such that

$$\mathbf{E}\chi_0(\xi\cap C) < \infty$$

for every compact  $C \subset \mathbb{R}^d$ , if and only if  $\Phi$  defined by (3.1) is positive and

(3.3) 
$$\int_{B\times B} \|x-y\|^{-d}\rho(dx\,dy) < \infty$$

for some open set B.

PROOF. The proof relies of several theorems that will appear later in this section. The first statement follows from Theorem 3.5 using the fact that the packing number  $P_t(X)$  of X is bounded by  $ct^{-d}$  for all sufficiently small *t*. For (ii), apply Theorem 3.9(ii) noticing that the imposed condition is equivalent to (3.24).

In the remainder of this section, one can find a quantification of this result [i.e., how the left-hand member of (3.3) controls the value of  $\mathbf{E}\chi_0(X \cap C)$ ] as well as generalisations for general metric spaces. The main argument used is a splitting method based on Theorem 2.6; the details are made clear in the proof of Theorem 3.3. Note that the packing number of the metric space appears as a crucial quantity to uncouple in this way the realisability problem; see Lemma 3.2.

3.2. *Moment conditions*. The family  $\mathcal{X}_k$  of all counting measures with total mass at most k on a compact space  $\mathbb{X}$  is compact. Thus, a measure  $\rho$  on  $\mathbb{X} \times \mathbb{X}$  is realisable as a point process with at most k points if (3.2) holds with  $\mathcal{X} = \mathcal{X}_k$ .

Assume that *Y* is a finite counting measure. For  $\alpha > 2$ , define

$$\chi_{\alpha}(Y) = Y(\mathbb{X})^{\alpha}, \qquad Y \in \mathcal{N}.$$

The finiteness of  $\mathbf{E}\chi_{\alpha}(\xi)$  amounts to the finiteness of the moment of order  $\alpha$  for the total mass of  $\xi$ . Since  $h \in \mathcal{C}_0$  is bounded by a constant c' and  $\alpha > 2$ , the family

$$\left\{Y \in \mathcal{N} : \chi_{\alpha}(Y) \le c + \mathsf{g}_{h}(Y)\right\} \subset \left\{Y \in \mathcal{N} : Y(\mathbb{X})^{\alpha} \le c + c'Y(\mathbb{X})^{2}\right\}$$

consists of counting measures with total masses bounded by a certain constant and, therefore, is compact in the space N. Hence,  $\chi_{\alpha}$  is a regularity modulus and so Theorem 2.6 yields the realisability condition

(3.4) 
$$\sup_{\mathsf{g}\in\mathsf{G},\mathsf{g}\leq\chi_{\alpha}}\Phi(\mathsf{g})<\infty$$

of  $\rho$  by a point process  $\xi$  whose total number of points has finite moment of order  $\alpha$ . Note that [13], Theorem 3.14, provides a variant of this result assuming the existence of the third factorial moment of the cardinality of  $\xi$  (i.e., with  $\alpha = 3$ ) and for the joint realisability of the intensity and the correlation measures. The condition of [13], Theorem 3.14 (reformulated for the correlation measure only) reads in our notation as  $c + \Phi(g_h) + br \ge 0$  whenever  $c + g_h + b\chi_3$  is nonnegative on  $\mathcal{N}$ . Noticing that  $b \ge 0$ , this is equivalent to the fact that  $c + \Phi(g_h) \le r$ whenever  $c + g_h \le \chi_3$ , being exactly (3.4). If  $\Theta$  is a group of continuous transformations acting on  $\mathbb{X}$  and  $\rho$  is  $\Theta$ -invariant, then the point process  $\xi$  can be chosen  $\Theta$ -stationary by Theorem 2.10(ii).

In order to handle possibly nonfinite point processes  $\xi$ , define

$$\chi_{\alpha,\beta}(Y) = \left(\sum_{x \in Y} \beta(x)\right)^{\alpha}, \qquad Y \in \mathcal{N},$$

for a lower semi-continuous strictly positive function  $\beta : \mathbb{X} \mapsto \mathbb{R}$  and  $\alpha > 2$ . By approximating  $\beta$  from below with compactly supported functions, it is easy to see that  $\chi_{\alpha,\beta}$  is a regularity modulus. By Theorem 2.6 and (2.2), for any given  $r \ge 0$ , there is a point process  $\xi$  with correlation measure  $\rho$  such that  $\mathbf{E}\chi_{\alpha,\beta}(\xi) \le r$  if and only if  $\rho$  satisfies

(3.5) 
$$\inf_{Y \in \mathcal{X}} [\chi_{\alpha,\beta}(Y) - g_h(Y)] + \int_{\mathbb{X} \times \mathbb{X}} h(x, y) \rho(dx \, dy) \le r, \qquad h \in \mathscr{C}_0$$

For  $\alpha = 3$ , condition (3.5) is a reformulation of [13], Theorem 3.17, meaning the positivity of  $\Phi$  on a family of positive polynomials that involve symmetric functions of the support points up to the third order. The realisability condition for  $\Theta$ -stationary random elements can be obtained by applying Theorem 2.10. 3.3. *Hardcore point processes on a compact space*. Assume that  $\mathbb{X}$  is a compact metric space with metric **d**. Let  $\mathcal{N}_{\varepsilon}$  be the family of  $\varepsilon$ -hard-core point sets in  $\mathbb{X}$  (including the empty set), that is, each  $Y \in \mathcal{N}_{\varepsilon}$  attaches unit masses to distinct points with pairwise distances at least  $\varepsilon$  with a fixed  $\varepsilon > 0$ . In this case, no multiple points are allowed, that is,  $\mathcal{N}_{\varepsilon} \subset \mathcal{N}_{0}$ .

According to [7, 10], a subset  $\mathcal{X}$  of simple counting measures  $\mathcal{N}_0$  is relatively compact if and only if  $\sup\{Y(K): Y \in \mathcal{X}\}$  is finite and the infimum over  $Y \in \mathcal{X}$ of the minimal distance between two points in  $Y \cap K$  is strictly positive for each compact set  $K \subset \mathbb{X}$ . The hard-core condition yields that the number of points in any compact set is uniformly bounded, and so  $\mathcal{N}_{\varepsilon}$  is indeed compact. By Corollary 2.7,  $\rho$  is realisable as the correlation measure of an  $\varepsilon$ -hard-core point process with given  $\varepsilon > 0$  if and only if

(3.6) 
$$\Phi(\mathbf{g}_h) \ge \inf_{Y \in \mathcal{N}_{\varepsilon}} \mathbf{g}_h(Y)$$

for all  $h \in \mathcal{C}_0$ . This result is formulated in [13], Theorem 3.4, which essentially reduces to the positivity of  $\Phi$  over the family  $c + g_h$  (in our setting).

In this paper, we assume that the hardcore distance is not predetermined and the point process takes realisations from  $\bigcup_{\varepsilon>0} \mathcal{N}_{\varepsilon}$ , which coincides with  $\mathcal{N}_0$  in case of compact  $\mathcal{X}$ . Note that (3.6) is stronger than the positivity of  $\Phi$  on functions  $g_h$  defined on the whole family  $\mathcal{N}_0$  and formulated as

(3.7) 
$$\Phi(\mathbf{g}_h) \ge \inf_{Y \in \mathcal{N}_0} \mathbf{g}_h(Y), \qquad h \in \mathscr{C}_0.$$

If X does not have isolated points, then the infimum in (3.7) can be taken over N. This is seen by approximating a multiple atom with a sequence of simple counting measures supported by points converging to the atom's location.

In the following, we use the (hard-core) regularity modulus of the form

$$\chi_{\psi}^{\mathrm{hc}}(Y) = \sum_{x_i, x_j \in Y, i \neq j} \psi(\mathbf{d}(x_i, x_j)), \qquad Y \in \mathcal{N}_0,$$

where  $\psi : (0, \infty) \mapsto [0, \infty]$  is a monotone decreasing right-continuous function, such that  $\psi(t) \to \infty$  as  $t \downarrow 0$ . The compactness of X and the lower semi-continuity of  $\psi$  imply that  $\chi_{\psi}^{hc}$  is lower semi-continuous on  $\mathcal{N}_0$ . As shown below  $\chi_{\psi}^{hc}$  is a regularity modulus if  $\psi$  grows sufficiently fast at zero.

Let  $P_t(\mathbb{X})$  be the *packing number* of  $\mathbb{X}$ , that is, the maximum number of points in  $\mathbb{X}$  with pairwise distances exceeding t, see [20], page 78. It is convenient to define the packing number at t = 0 as  $P_0(\mathbb{X}) = \infty$  if  $\mathbb{X}$  is infinite and otherwise let  $P_0(\mathbb{X})$  be the cardinality of  $\mathbb{X}$ .

LEMMA 3.2. Function 
$$\chi_{\psi}^{\text{hc}}$$
 is a regularity modulus on  $\mathcal{N}_0$  if  
(3.8)  $\psi(t)/P_t(\mathbb{X}) \to \infty$  as  $t \downarrow 0$ .

PROOF. In view of the compactness of  $\mathbb{X}$ , it is possible to bound  $h \in \mathscr{C}_0$  by a constant  $\lambda$ , so that  $\chi_{\psi}^{hc}$  is a regularity modulus if

$$\mathcal{H}_{\lambda} = \left\{ Y \in \mathcal{N}_0 : \chi_{\psi}^{\text{hc}}(Y) \le \lambda Y(\mathbb{X})^2 \right\}$$

is compact in  $\mathcal{N}_0$  for each  $\lambda > 0$ . For this, it suffices to show that the total mass of all  $Y \in \mathcal{H}_{\lambda}$  is bounded by a fixed number and  $\mathcal{H}_{\lambda} \subset \mathcal{N}_{\varepsilon}$  for some  $\varepsilon > 0$ .

Let  $\gamma_t(n)$  be the minimal number of pairs  $(x_i, x_j)$  with  $i \neq j$ , such that  $x_i, x_j \in Y$  and  $\mathbf{d}(x_i, x_j) \leq t$  over all counting measures *Y* of total mass *n*.

Take *t* such that  $\psi(t)/P_t(\mathbb{X}) > \lambda$ . If  $Y(\mathbb{X}) \ge n$ , then

$$\chi_{\psi}^{\mathrm{hc}}(Y) \geq \sum_{x_i, x_j \in Y, i \neq j} \psi(t) \mathbb{1}_{\mathbf{d}(x_i, x_j) \leq t} \geq \gamma_t(n) \psi(t).$$

Therefore,

$$\mathcal{H}_{\lambda} \subset \left\{ Y : n^{-2} \gamma_t(n) \psi(t) \le \lambda \right\}$$

consists of *Y* with total mass uniformly bounded by fixed number  $n_{\lambda}$ . Indeed, by Lemma A.1,

$$\lim_{n \to \infty} n^{-2} \gamma_t(n) \ge \lim_{n \to \infty} n^{-2} n \left( \frac{n}{P_t(\mathbb{X})} - 1 \right) = P_t(\mathbb{X})^{-1}$$

Choose  $\varepsilon > 0$  so that  $\psi(t) \ge \lambda n_{\lambda}^2$  for  $t \le \varepsilon$ . For  $Y \in \mathcal{H}_{\lambda}$  and any  $x_i, x_j \in Y$ ,

$$\psi(\mathbf{d}(x_i, x_j)) \leq \chi_{\psi}^{\mathrm{hc}}(Y) \leq \lambda n_{\lambda}^2$$

whence  $\mathbf{d}(x_i, x_j) \ge \varepsilon$ . Thus,  $\mathcal{H}_{\lambda} \subset \mathcal{N}_{\varepsilon}$ , so  $\mathcal{H}_{\lambda}$  is relatively compact.  $\Box$ 

The following theorem shows that the realisability condition can be split into the positivity condition (3.7) on the linear functional  $\Phi$  and the regularity condition (3.9) on the correlation measure, so that the latter can be easily checked. Such a split is possible because the regularity modulus  $\chi_{\psi}^{hc}$  can be approximated by functions from G.

THEOREM 3.3. A locally finite measure  $\rho$  on  $\mathbb{X} \times \mathbb{X}$  is the correlation measure of a simple point process  $\xi$  such that  $\mathbf{E}\chi_{\psi}^{hc}(\xi) \leq r$  for some  $r \geq 0$  with  $\psi$  satisfying (3.8) if and only if (3.7) holds and

(3.9) 
$$\int_{\mathbb{X}\times\mathbb{X}}\psi(\mathbf{d}(x,y))\rho(dx\,dy) \leq r.$$

PROOF. Necessity. The definition of the correlation measure implies that

$$\int_{\mathbb{X}\times\mathbb{X}}\psi(\mathbf{d}(x,y))\rho(dx\,dy) = \mathbf{E}\chi_{\psi}^{\mathrm{hc}}(\xi) \leq r.$$

Sufficiency. First assume that  $\psi$  only takes finite values. The proof consists of checking (2.2), which is equivalent to (2.5).

For each family of positive numbers  $\{t_g, g \in G\}$ ,

(3.10) 
$$\sup_{\mathsf{g}\in\mathsf{G}}\inf_{Y\in\mathcal{N}_0}[\chi(Y)-\mathsf{g}(Y)]+\Phi(\mathsf{g})\leq \sup_{\mathsf{g}\in\mathsf{G}}\inf_{Y\in\mathcal{N}_{l_{\mathsf{g}}}}[\chi(Y)-\mathsf{g}(Y)]+\Phi(\mathsf{g}).$$

The crucial step of the proof consists in the careful choice of  $t_{g} > 0$ .

Fix  $g \in G$ . For t > 0, define  $\psi_t(s) = \psi(\max(t, s))$ ,  $s \ge 0$ . Since any  $Y \in \mathcal{N}_t$  does not contain any two points at distance less than t,  $\chi(Y) = g_{\psi_t}(X)$ . Therefore,

(3.11) 
$$\inf_{Y \in \mathcal{N}_t} [\chi(Y) - \mathsf{g}(Y)] = \inf_{Y \in \mathcal{N}_t} (\mathsf{g}_{\psi_t} - \mathsf{g})(Y).$$

Our aim is to prove that

(3.12) 
$$\inf_{Y \in \mathcal{N}_t} (\mathbf{g}_{\psi_t} - \mathbf{g})(Y) = \inf_{Y \in \mathcal{N}_0} (\mathbf{g}_{\psi_t} - \mathbf{g})(Y),$$

because then, since  $g_{\psi_t} \in G$ , the positivity of  $\Phi$  on G yields that (3.11) is not greater than  $\Phi(g_{\psi_t} - g)$ . Thus, (3.10) is bounded above by

$$\sup_{\mathsf{g}\in\mathsf{G}}\Phi(\mathsf{g}_{\psi_t}-\mathsf{g})+\Phi(\mathsf{g})\leq\sup_t\Phi(\mathsf{g}_{\psi_t})=\int_{\mathbb{X}\times\mathbb{X}}\psi(\mathsf{d}(x,y))\rho(dx\,dy)$$

by the monotone convergence theorem.

The proof of (3.12) relies on the proper choice for *t* (depending on g). Assume without loss of generality  $g = g_h$  for  $h \in \mathcal{C}_0$  with absolute value bounded by  $\lambda > 0$ . By (3.8), there exists  $t_0$  such that  $\psi(t_0)/P_{t_0}(\mathbb{X}) \ge \lambda + 1$ . By Lemma A.1, there is  $n_0$  such that for all *Y* with mass  $n \ge n_0$ , the number of pairs of points of *Y* at distance at most  $t_0$  satisfies

$$\gamma_{t_0}(n) \ge n^2 \frac{1}{P_{t_0}(\mathbb{X})}.$$

Choose  $t \le t_0$  so that  $\psi(t) > \lambda n_0^2$  and consider any  $Y \in \mathcal{N}_0 \setminus \mathcal{N}_t$ . If  $Y(\mathbb{X}) \le n_0$ , then

$$\mathsf{g}_{\psi_t}(Y) \ge \psi(t) > \lambda n_0^2 \ge \mathsf{g}_h(Y),$$

while if  $Y(\mathbb{X}) > n_0$ , then

$$\mathsf{g}_{\psi_t}(Y) - \mathsf{g}_h(Y) \ge \mathsf{g}_{\psi_{t_0}}(Y) - \mathsf{g}_h(Y) \ge \psi(t_0)\gamma_{t_0}(Y) - \lambda n^2 > 0.$$

Thus for  $Y \notin \mathcal{N}_t$ , we have  $g_{\psi_t}(Y) - g_h(Y) > 0$ . Therefore, the infimum of  $g_{\psi_t} - g_h$ , which is nonpositive because zero is obtained for  $Y = \emptyset$ , is reached on  $\mathcal{N}_t$ , and (3.12) is proved.

Now assume that  $\psi(t)$  is infinite for  $t \in [0, \delta)$  and finite on  $(\delta, \infty)$  with  $\delta > 0$ . If  $\psi(t) \to \infty$  as  $t \downarrow \delta$ , then the above arguments apply with  $t_0 > \delta$  chosen such that  $\psi(t_0)/P_{\delta}(\mathbb{X}) > \lambda$ .

Assume that  $\psi(\delta)$  is finite. Let  $\psi_0(t)$  be a function satisfying (3.8) and finite for all t > 0, for example,  $\psi_0(t) = t^{-1} P_t(\mathbb{X})$ . Define  $\psi^*(t) = \psi(t)$  for  $t \ge \delta$  and let  $\psi^*(t) = \psi_0(t) + a$  for  $t \in (0, \delta)$  with a sufficiently large *a*, so that  $\psi^*$  is monotone

right-continuous, and  $\chi_{\psi^*}^{hc}$  is a regularity modulus. Applying the previous arguments to  $\psi^*$  yields that there exists a point process  $\xi$  such that  $\mathbf{E}\chi_{\psi^*}^{hc}(\xi) \leq r$ . Since  $r < \infty$ ,  $\rho$  vanishes on  $\{(x, y) : \mathbf{d}(x, y) < \delta\}$ , and so  $\mathbf{E}\chi_{\psi^*}^{hc}(\xi) = \mathbf{E}\chi_{\psi}^{hc}(\xi) \leq r$ .  $\Box$ 

The following result is obtained by letting  $\psi$  be infinite on  $[0, \varepsilon)$  and otherwise setting it to zero.

COROLLARY 3.4. A measure  $\rho$  on  $\mathbb{X} \times \mathbb{X}$  is the correlation measure of a point process  $\xi$  with  $\xi \in \mathcal{N}_{\varepsilon}$  a.s. if and only if (3.7) holds and  $\rho(\{(x, y) : \mathbf{d}(x, y) < \varepsilon\}) = 0$ .

The following result yields a direct realisability condition for  $\rho$  without mentioning a regularity modulus.

THEOREM 3.5. Let  $\rho$  be a locally finite measure on  $\mathbb{X} \times \mathbb{X}$ , and fix any  $r \ge 0$ . Then there exists, for every r' > r, a simple point process  $\xi$  with correlation measure  $\rho$ , such that

(3.13) 
$$\mathbf{E}\sum_{x_i, x_j \in \xi, i \neq j} P_{\mathbf{d}(x_i, x_j)}(\mathbb{X}) \leq r',$$

if and only if (3.7) holds and

(3.14) 
$$\int_{\mathbb{X}\times\mathbb{X}} P_{\mathbf{d}(x,y)}(\mathbb{X})\rho(dx\,dy) \le r.$$

PROOF. Necessity. Call  $h_t(x, y) = \min(t, P_{\mathbf{d}(x, y)}(\mathbb{X}))$  for  $x \neq y \in \mathbb{X}$  and t > 0. Assume that  $\xi$  realises  $\rho$  and satisfies (3.13). The monotone convergence theorem yields that

$$\int_{\mathbb{X}\times\mathbb{X}} P_{\mathbf{d}(x,y)}(\mathbb{X})\rho(dx\,dy) = \lim_{t\to\infty} \mathbf{E}\mathbf{g}_{h_t}(\xi) \le r'$$

for every r' > r, whence (3.14) holds.

*Sufficiency*. Define a measure on  $\mathbb{R}_+$  by

$$\rho'([a,b)) = \rho(\{(x, y) \in \mathbb{X} \times \mathbb{X} : a \le \mathbf{d}(x, y) < b\}).$$

Fubini's theorem yields that

$$r = \int_{\mathbb{R}_+} P_t(\mathbb{X}) \rho'(dt).$$

Let  $\{t_k, k \ge 1\}$  be a strictly decreasing sequence of numbers such that

$$\int_{[0,t_k)} P_t(\mathbb{X})\rho'(dt) \leq 2^{-k}.$$

For  $m \ge 1$ , the function

$$\psi_m(t) = \begin{cases} k P_t(\mathbb{X}), & \text{if } t_{k+1} \le t < t_k < t_m, k \ge 1, \\ P_t(\mathbb{X}), & \text{if } t \ge t_m \end{cases}$$

is monotone right-continuous and satisfies  $\psi_m(t)/P_t(\mathbb{X}) \to \infty$  as  $t \to 0$ . Then

$$\begin{split} \int_{\mathbb{X}\times\mathbb{X}} \psi_m (\mathbf{d}(x, y)) \rho(dx \, dy) &= \int_{\mathbb{R}_+} \psi_m(t) \rho'(dt) \\ &\leq \int_{\mathbb{R}_+} P_t(\mathbb{X}) \rho'(dt) + \sum_{k \ge m} k 2^{-k} \le r + \sum_{k \ge m} k 2^{-k}. \end{split}$$

By Theorem 3.3, choosing *m* sufficiently large yields the realisability of  $\rho$  by a point process  $\xi$  satisfying

$$\mathbf{E}\sum_{x_i, x_j \in \xi, i \neq j} P_{\mathbf{d}(x_i, x_j)}(\mathbb{X}) \le \mathbf{E}\chi_{\psi_m}^{\mathrm{hc}}(\xi) \le r + \sum_{k \ge m} k2^{-k} < r'.$$

REMARK 3.6. Let  $\Theta$  be a group of continuous transformations on  $\mathbb{X}$  that leave  $\rho$  invariant, that is,  $\rho(\theta A \times \theta B) = \rho(A \times B)$  for all  $\theta \in \Theta$  and Borel *A*, *B*. Since the regularity modulus  $\chi_{\psi}^{hc}$  can be approximated from below by a sequence of functions from G, Theorem 2.10(i) is applicable and so the corresponding point process  $\xi$  in Theorems 3.3, 3.5 and Corollary 3.4 can be chosen  $\Theta$ -stationary. If  $\Theta$  consists of isometric transformations, then Theorem 2.10(ii) is also applicable.

3.4. Noncompact case and stationarity. Assume that  $\mathbb{X} = \mathbb{R}^d$  and  $\mathbf{d}(x, y) = ||x - y||$  is the Euclidean metric. Let  $\psi$  be a positive right-continuous monotone function on  $\mathbb{R}_+$  such that  $\psi(t)t^d \to \infty$  as  $t \to 0$ . Denote by  $B_n$  the open ball of radius *n* centred at 0. Given a known bound for the packing number in the Euclidean space ([20], page 78, Lemma 3.2) implies that  $\chi_{\psi}^{\text{hc}}$  is a regularity modulus on every  $B_n$ ,  $n \ge 1$ . Define

(3.15) 
$$\chi_{\beta\psi}^{\mathrm{hc}}(Y) = \sum_{x_i, x_j \in Y, i \neq j} \beta(x_i, x_j) \psi\big( \|x_i - x_j\|\big)$$

for a bounded lower semi-continuous strictly positive on  $\mathbb{R}^d \times \mathbb{R}^d$  function  $\beta$ .

THEOREM 3.7. Let  $\rho$  be a locally finite measure on  $\mathbb{R}^d \times \mathbb{R}^d$ .

(i) The measure  $\rho$  is realisable as the correlation measure of a point process  $\xi$  that satisfies  $\mathbf{E}\chi^{hc}_{\beta\psi}(\xi) \leq r$  if and only if (3.7) holds and

(3.16) 
$$\int_{\mathbb{R}^d \times \mathbb{R}^d} \beta(x, y) \psi(\|x - y\|) \rho(dx \, dy) \le r.$$

(ii) Fix  $r \ge 0$ , let

(3.17) 
$$r_n = \int_{B_n \times B_n} \|x - y\|^{-d} \rho(dx \, dy), \qquad n \ge 1,$$

and let  $\{\beta_n, n \ge 1\}$  be a sequence of nonincreasing numbers converging to 0. Then the following assertions are equivalent.

(a) Equation (3.7) holds and

(3.18) 
$$\sum_{n\geq 1}\beta_n(r_{n+1}-r_n)\leq r<\infty.$$

in particular every  $r_n$ ,  $n \ge 1$ , is finite.

(b) For every r' > r, there exists  $\xi$  with correlation measure  $\rho$  and such that

(3.19) 
$$\sum_{n\geq 1} (\beta_n - \beta_{n+1}) \mathbf{E} \sum_{x_i, x_j \in B_n, i \neq j} \|x_i - x_j\|^{-d} \le r'.$$

**PROOF.** Sufficiency. (i) The function  $\chi^{hc}_{\beta\psi}$  is a regularity modulus on  $\mathcal{N}_0$ , since

$$\mathcal{H}_{c,h} = \{ Y \in \mathcal{X} : \chi_{\beta\psi}^{\mathrm{hc}}(Y) \le c + \mathsf{g}_h(Y) \}, \qquad c \in \mathbb{R}, h \in \mathscr{C}_0,$$

is compact in  $\mathcal{N}_0$ . This follows from Lemma 3.2, which yields the compactness of the restriction of *Y* from  $\mathcal{H}_{c,h}$  onto any compact set *C*. Indeed, this family of restricted counting measures coincides with the family of simple counting measures supported by *C* such that  $\chi_{\psi}^{\text{hc}}(Y) \leq c/m + g_{h/m}(Y)$ , where m > 0 is a lower bound of  $\beta(x, y)$  for  $x, y \in C$ .

In order to apply Theorem 2.6 with the regularity modulus (3.15) and in view of (2.5) it suffices to show that

(3.20) 
$$\inf_{Y \in \mathcal{N}_0} \left[ \chi_{\beta \psi}^{\text{hc}}(Y) - \mathsf{g}_h(Y) \right] + \int_{\mathbb{R}^d \times \mathbb{R}^d} h(x, y) \rho(dx \, dy) \le r$$

for all  $h \in \mathcal{C}_0$ . Assume that *h* is supported by a subset of  $B_n \times B_n$  for some  $n \ge 1$ . Then (3.20) holds by the same reasoning as in the proof of Theorem 3.3 applied to the compact space  $B_n$ . (One might first consider only  $Y \subset B_n$ , and then note that the infimum over all  $Y \in \mathcal{N}_0$  is necessarily smaller.) By Theorem 2.6, (3.16) implies the existence of a point process  $\xi$  with correlation measure  $\rho$  that satisfies  $\mathbf{E}\chi^{hc}_{\beta\psi}(\xi) \le r$ .

(ii) Define  $\mathbb{Y}_n = (B_n \times B_n) \setminus (B_{n-1} \times B_{n-1}), n \ge 1$  (with  $B_0 = \emptyset$ ). For every  $n \ge 1$ , define the measure

$$\rho'_n([a,b)) = \rho(\{(x, y) \in B_n \times B_n : a \le ||x - y|| < b\}),$$

and let

$$t_k^n = \sup\left\{t > 0: \int_{[0,t]} s^{-d} \rho'_n(ds) \le 2^{-k}\right\}, \qquad k \ge 1.$$

134

Since  $\rho'_{n+1} \ge \rho'_n$  for every  $n \ge 1$ , for every  $k, n \ge 1$  we have  $t_k^{n+1} \le t_k^n$ . Let  $\{m_n, n \ge 1\}$  be a nondecreasing sequence of positive integers so that

(3.21) 
$$\sum_{n\geq 1} \beta_n \left( r_n - r_{n-1} + \sum_{k\geq m_n} k 2^{-k} \right) \leq r'.$$

Now define

$$\psi_n(t) = \begin{cases} kt^{-d}, & \text{if } t_{k+1}^n \le t < t_k^n < t_{m_n}^n, \\ t^{-d}, & \text{if } t \ge t_{m_n}^n. \end{cases}$$

Since  $m_n \le m_{n+1}$ ,  $\psi_{n+1} \le \psi_n$  for every  $n \ge 1$ . Function  $\psi_n$  satisfies  $\psi_n(t)t^d \to \infty$  as  $t \to 0$ , whence, for every  $n \ge 1$ ,  $\chi_{\psi_n}^{hc}$  is a regularity modulus on counting measures supported by  $B_n$  and

$$\int_{\mathbb{Y}_n} \psi_n \big( \|x - y\| \big) \rho(dx \, dy) \le \int_{\mathbb{R}_+} \psi_n(t) \rho_n''(dt),$$

where

$$\rho_n''([a,b)) = \rho(\{(x, y) \in \mathbb{Y}_n : a \le ||x - y|| < b\}) \le \rho_n'([a,b)).$$

Then

$$\int_{\mathbb{Y}_n} \psi_n(\|x-y\|) \rho(dx\,dy) \leq \int_{\mathbb{R}_+} t^{-d} \rho_n''(dt) + \int_{t \leq t_{m_n}} \psi_n(t) \rho_n''(dt).$$

We have

$$\int_{\mathbb{R}_{+}} t^{-d} \rho_{n}''(dt) = \int_{(B_{n} \times B_{n}) \setminus (B_{n-1} \times B_{n-1})} \|x - y\|^{-d} \rho(dx \, dy) = r_{n} - r_{n-1}$$

and

$$\int_{t\leq t_{m_n}}\psi_n(t)\rho_n''(dt)\leq \int_{t\leq t_{m_n}}\psi_n(t)\rho_n'(dt)\leq \sum_{k\geq m_n}k2^{-k},$$

whence

(3.22) 
$$\int_{\mathbb{Y}_n} \psi_n (\|x - y\|) \rho(dx \, dy) \leq r_n - r_{n-1} + \sum_{k \geq m_n} k 2^{-k}.$$

Define  $\psi(x, y) = \psi_n(||x - y||)$  for  $x, y \in \mathbb{Y}_n$ . Since  $\psi_{n+1} \leq \psi_n$  and functions  $\psi_n, n \geq 1$ , are lower semi-continuous, the function  $\psi$  is lower semi-continuous on  $\mathbb{R}^d \times \mathbb{R}^d$ . Define  $\beta(x, y) = \beta_n$  on  $\mathbb{Y}_n$ . Since  $\beta_n, n \geq 1$ , decrease,  $\beta$  is a lower semi-continuous function on  $\mathbb{R}^d \times \mathbb{R}^d$ . Since  $\psi_n \leq \psi_k$  for every  $k \leq n$ , the restriction of  $\chi_{\beta\psi}^{hc}$  onto sets  $Y \subset B_n$  is larger than  $\chi_{\beta_n\psi_n}^{hc}$ , whence  $\chi_{\beta\psi}^{hc}$  is a regularity modulus on  $\mathcal{N}_0$ . By Theorem 2.6,  $\Phi$  is realised by a point process  $\xi$  satisfying

$$\mathbf{E}\chi_{\beta\psi}^{\mathrm{hc}}(\xi) \leq \int_{\mathbb{R}^d \times \mathbb{R}^d} \beta(x, y)\psi(x, y)\rho(dx\,dy) = \sum_{n \geq 1} \beta_n \int_{\mathbb{Y}_n} \psi_n(\|x - y\|)\rho(dx\,dy).$$

Since  $t^{-d} \leq \psi_n(t)$  for each *n* and t > 0,

$$\mathbf{E}\chi_{\beta\psi}^{\mathrm{hc}}(\xi) = \lim_{m \to \infty} \mathbf{E}\sum_{n=1}^{m} \beta_n \left( \chi_{\psi_n}^{\mathrm{hc}}(\xi \cap B_n) - \chi_{\psi_n}^{\mathrm{hc}}(\xi \cap B_{n-1}) \right)$$
$$\geq \lim_{m \to \infty} \mathbf{E}\sum_{n=1}^{m} (\beta_n - \beta_{n+1}) \chi_{\psi_n}^{\mathrm{hc}}(\xi \cap B_n) \right)$$
$$\geq \sum_{n \ge 1} (\beta_n - \beta_{n+1}) \mathbf{E}\sum_{i \ne j, x_i, x_j \in B_n} \|x_i - x_j\|^{-d}.$$

Using successively (3.22) and (3.21),

$$\sum_{n\geq 1}\beta_n\int_{\mathbb{Y}_n}\psi_n(\|x-y\|)\rho(dx\,dy)\leq \sum_{n\geq 1}\beta_n\left(r_n-r_{n-1}+\sum_{k\geq m_n}k2^{-k}\right)\leq r',$$

we arrive at (3.19).

*Necessity.* For (ii), remark first that for  $x \neq y \in \mathbb{R}^d$ 

$$\beta(x, y) \|x - y\|^{-d} = \sum_{n \ge 1} (\beta_n - \beta_{n+1}) \mathbb{1}_{x, y \in B_n} \|x - y\|^{-d},$$

where  $\beta$  is defined in the sufficiency part of the proof. The function  $\beta \psi$  in (i) and the function  $(x, y) \mapsto \beta(x, y) ||x - y||^{-d}$  in (ii) are lower semi-continuous and can therefore be approximated from below by compactly supported continuous functions. The rest follows from the monotone convergence theorem similarly to the proof of necessity in Theorem 3.5.  $\Box$ 

REMARK 3.8. Remark for point (ii) that if each  $r_n$ ,  $n \ge 1$ , is finite, there always exists a sequence  $\{\beta_n\}$  of sufficiently small numbers such that the right-hand side of (3.18) is finite.

If the distribution of point process  $\xi$  is invariant with respect to the group  $\Theta$  of translations of  $\mathbb{R}^d$ , then  $\xi$  is called *stationary*. Its correlation measure  $\rho$  is translation invariant, that is,  $\rho((A + x) \times (B + x)) = \rho(A \times B)$  for all  $x \in \mathbb{R}^d$  and so

(3.23) 
$$\rho(A \times B) = \lambda^2 \int_A \int_{\mathbb{R}^d} \mathbb{1}_{x+y \in B} \bar{\rho}(dy) \, dx,$$

where  $\lambda$  is the intensity of  $\xi$  and  $\bar{\rho}$  is a measure on  $\mathbb{R}^d$  called the *reduced* correlation measure; see [25], page 76.

THEOREM 3.9. Let  $\bar{\rho}$  be a locally finite measure on  $\mathbb{R}^d$ , let  $\beta$  be a bounded lower semi-continuous strictly positive function on  $\mathbb{R}^d$  satisfying

$$\bar{\beta}(y) = \int_{\mathbb{R}^d} \beta(x, x+y) \, dx < \infty, \qquad y \in \mathbb{R}^d,$$

and let  $\psi$  be a monotone decreasing nonnegative function such that  $t^d \psi(t) \to \infty$ .

(i)  $\bar{\rho}$  is the reduced correlation measure of a stationary point process  $\xi$  that satisfies  $\mathbf{E}\chi^{hc}_{\beta\psi}(\xi) \leq r$  if and only if (3.7) holds and

$$\int_{\mathbb{R}^d} \bar{\beta}(y) \psi\big(\|y\|\big) \bar{\rho}(dy) \le r.$$

(ii)  $\bar{\rho}$  is realisable as the reduced correlation measure of a stationary point process  $\xi$  that satisfies (3.19) for some sequence  $\{\beta_n, n \ge 1\}$  if and only if

(3.24) 
$$\int_B \|y\|^{-d} \bar{\rho}(dy) < \infty$$

for some open ball *B* containing the origin. If  $\int_{\mathbb{R}^d} ||y||^{-d} \bar{\rho}(dy)$  is finite, it is possible to let  $\beta_n = n^{-d-\delta}$ ,  $n \ge 1$ , for any fixed  $\delta > 0$ .

PROOF. It suffices to use (3.23) to confirm the conditions imposed in Theorem 3.7, see also Remark 3.8. In order to show that  $\xi$  can be chosen stationary, note that  $\chi^{hc}_{\beta\psi}$  can be pointwisely approximated from below by a monotone sequence of functions from G, so Theorem 2.10(i) applies.

3.5. *Joint realisability of the intensity and correlation*. Recall that the intensity measure  $\rho_1$  of a point process  $\xi$  is defined from

$$\mathbf{E}\sum_{x_i\in\xi}h(x_i)=\int h(x)\rho_1(dx),\qquad h\in\mathscr{C}_{\mathbf{0},1},$$

where  $\mathscr{C}_{0,1}$  is the family of continuous functions on  $\mathbb{X}$  with compact support. A pair  $(\rho_1, \rho)$  of locally finite nonnegative measures on  $\mathbb{X}$  and  $\mathbb{X} \times \mathbb{X}$ , respectively, is said to be jointly realisable if there exists a point process  $\xi$  with intensity measure  $\rho_1$  and correlation measure  $\rho$ .

Let G<sub>1</sub> be the vector space formed by constants and functions

$$\mathsf{g}_{h_1,h}(Y) = \sum_{x \in Y} h(x) + \mathsf{g}_h(Y), \qquad Y \in \mathcal{N},$$

for  $h_1 \in \mathcal{C}_{0,1}$  and  $h \in \mathcal{C}_0$ . It is easy to see that Assumption 2.4 is verified in this case. The pair  $(\rho_1, \rho)$  yields a linear functional

(3.25) 
$$\Phi(\mathbf{g}_{h_1,h}) = \int_{\mathbb{X}} h_1(x)\rho_1(dx) + \int_{\mathbb{X}\times\mathbb{X}} h(x,y)\rho(dx\,dy).$$

The realisability of  $\Phi$  by a point process  $\xi$  means that  $\Phi(g_{h_1,h}) = \mathbf{E}g_{h_1,h}(\xi)$ . Functional  $\Phi$  is positive on  $G_1$  if and only if

(3.26) 
$$\Phi(\mathbf{g}_{h_1,h}) \ge \inf_{Y \in \mathcal{X}} \mathbf{g}_{h_1,h}(Y), \qquad h_1 \in \mathscr{C}_{0,1}, h \in \mathscr{C}_0.$$

Similar arguments as before apply and yield the joint realisability conditions. Consider the special case of stationary processes in  $\mathbb{X} = \mathbb{R}^d$  with the reduced correlation measure  $\bar{\rho}$  [see (3.23)] and intensity  $\rho_1(dx) = \lambda dx$  proportional to the Lebesgue measure.

THEOREM 3.10. Let  $\lambda$  be a constant, and let  $\bar{\rho}$  be a locally finite measure of  $\mathbb{R}^d$ . Then there is a stationary point process  $\xi$  with intensity  $\rho_1(dx) = \lambda dx$  and reduced correlation measure  $\bar{\rho}$  if  $\Phi$  given by (3.25) satisfies (3.26) with  $\mathcal{X} = \mathcal{N}_0$ and

$$\int_B \|z\|^{-d}\bar{\rho}(dz) < \infty$$

for some open set B containing the origin.

PROOF. It suffices to note that  $g_{h_1,h}$  is dominated by  $cg_h$  for a constant c and follow the proof of (ii) in Theorem 3.7. The condition on  $\bar{\rho}$  follows from (3.17) and (3.23).

4. Realisability of covering probabilities for random sets. The nature of realisability problem changes with the choice of the family of subsets of a carrier space X taken as possible values for a random set. We start with the case when a random set is allowed to be any subset of X, where realisability results are available under minimal conditions, while the obtained random set lacks properties and might even not be measurable. In the remainder of this section, we treat the case of random closed sets, a classical setting in stochastic geometry. Some examples of possible regularity moduli are presented, along with the corresponding realisability results that resemble those of [12] in the point processes setting. The framework of random measurable sets with finite perimeter (in the variational sense), developed in the forthcoming paper [6], provides a compromise between regularity of the random set and the explicitness of conditions.

4.1. Random binary functions. Let  $\mathcal{X}$  be the family of all subsets of  $\mathbb{X}$  identified with their indicator functions. Endow  $\mathcal{X}$  with the topology of pointwise convergence and the corresponding  $\sigma$ -algebra. Since  $\mathcal{X}$  is compact, Corollary 2.7 yields the following result.

THEOREM 4.1. Let G be a vector space that consists of continuous functions on  $\mathcal{X}$  and includes constants, and let  $\Phi$  be a map from G to  $\mathbb{R}$ . Then there exists a random indicator function  $\xi$ , such that  $\Phi(g) = Eg(\xi)$  for all  $g \in G$  if and only if  $\Phi$  is a linear positive functional on G and  $\Phi(1) = 1$ .

The key issue in applying Theorem 4.1 is the choice of the space G.

EXAMPLE 4.2 (One-point covering function). Let G be generated by constants *c* and one-point indicator functions  $g_x(F) = \mathbb{1}_{x \in F}$ ,  $F \in \mathcal{X}$ , for  $x \in \mathbb{X}$ . The positivity of a linear functional  $\Phi: G \mapsto \mathbb{R}$  together with  $\Phi(1) = 1$  means that  $\Phi(g_x) \in [0, 1]$  for all  $x \in \mathbb{X}$ . Thus, a function  $p_x = \Phi(g_x)$  is a one-point covering function  $\mathbf{P}\{x \in \xi\}$  for a random set  $\xi$  if and only if  $p_x$  takes values in [0, 1]. Compare with Theorem 1.1, where the extra upper semi-continuity condition ensures that the corresponding random binary function is upper semi-continuous and so  $\xi$  is a random *closed* set.

EXAMPLE 4.3 (Covariances of random sets). Consider vector space G generated by constants and functions  $g_{x,y}(F) = \mathbb{1}_{x,y\in F}$  for  $x, y \in \mathbb{X}$ . The values of a linear functional  $\Phi$  on G are determined by  $p_{x,y} = \Phi(g_{x,y}), x, y \in \mathbb{X}$ . By (2.1),  $\Phi$  is positive on G if and only if

(4.1) 
$$\sum_{ij=1}^{n} a_{ij} p_{x_i, x_j} \ge \inf_{F \subset \mathbb{X}} \sum_{ij=1}^{n} a_{ij} \mathbb{1}_{x_i, x_j \in F}$$

for all  $n \ge 1$  and all matrices  $(a_{ij})_{ij=1}^n$ . In particular, if  $a_{ij} = a_i a_j$ , then (4.1) implies the nonnegative definiteness of  $p_{x,y}$ ,  $x, y \in \mathbb{X}$ . Note that the one-point covering probabilities are specified if  $p_{x,y}$  are given.

4.2. The closedness condition. A random closed set  $\xi$  in a locally compact metric space  $\mathbb{X}$  is a random element that takes values in the family  $\mathcal{X} = \mathcal{F}$  of closed subsets of  $\mathbb{X}$  equipped with the  $\sigma$ -algebra (called the Effros  $\sigma$ -algebra) generated by families  $\{F \in \mathcal{F} : F \cap K \neq \emptyset\}$  for all compact sets K. The distribution of a random closed set  $\xi$  is uniquely determined by its *capacity functional* 

$$T(K) = \mathbf{P}\{\xi \cap K \neq \emptyset\}$$

for all K from the family of all compact sets in X; see [18] and [22], Theorem 1.1.13.

Theorem 4.1 ensures only the existence of a binary stochastic process with given marginal distributions up to a certain order. However, it is not guaranteed that the constructed stochastic process has upper semi-continuous realisations, which should be the case if this process is the indicator of a random *closed* set in a topological space X. If the carrier space X is finite (more generally, has a discrete topology), then this problem is avoided, since each random set is closed. Furthermore, the closedness issue can be settled in the following special case of two-point probabilities in the product form (and can be generalised for multi-point covering probabilities). The following result implies, in particular, that the random indicator function from Example 1.2 does not correspond to a random closed set. It also illustrates regularity problems arising in realisability problems for random closed sets.

THEOREM 4.4. Assume that X is a separable space. A function

$$p_{x,y} = \begin{cases} p_x p_y, & \text{if } x \neq y, \\ p_x, & \text{if } x = y \end{cases}$$

is a two-point covering function of a random closed set if and only if  $p_x, x \in X$ , is an upper semi-continuous function with values in [0, 1] such that each point x with  $p_x \in (0, 1)$  has an open neighbourhood U such that  $p_y > 0$  only for at most a countable number of  $y \in U$  and the sum of  $p_y$  for  $y \in U$  is finite. PROOF. Sufficiency. Note that the set  $L = \{x : p_x = 1\}$  is closed by the upper semi-continuity of  $p_x$ . The separability of X and the condition of theorem imply that the set  $M = \{x : p_x \in (0, 1)\}$  is at most countable. The sufficiency is obtained by a direct construction of a random subset Z of M that contains each point x with probability  $p_x$  independently of all other points. It remains to show that the random set  $\xi = Z \cup L$  is closed. Consider  $x \in M$  and its neighbourhood from the condition of theorem. Since  $\sum p_y < \infty$ , only a finite number of y belong to Z and so they do not converge to x. Thus, x with probability zero appears as a limit of other points from  $\xi$  unless  $x \in L$  and so belongs to  $\xi$  almost surely.

*Necessity.* The function  $p_x = \mathbf{P}\{x \in \xi\}$  is upper semi-continuous, since  $\xi$  is a random closed set. The product form of the two-point covering function implies that the capacity functional on two-point set is given by

$$T(\{x, y\}) = p_x + p_y - p_x p_y.$$

The upper semi-continuity property of the capacity functional yields that

$$\limsup_{y \to x} T(\{x, y\}) \le p_x$$

while the monotonicity implies that  $T(\{x, y\}) \to p_x$  as  $y \to x$ . Thus,  $p_y(1 - p_x) \to 0$  as  $y \to x$  for all x. Unless  $p_x = 1$ , we have  $p_y \to 0$ .

Assume that  $p_x > 0$  and  $p_{x_n} > 0$ , where  $x_n \to x$  and  $x_n \neq x$  with  $\sum p_{x_n} = \infty$ . A variant of the lemma of Borel–Cantelli for pairwise independent events (see [5], Lemma 6.2.5) implies that almost surely infinitely many points  $x_n$  belong to  $\xi$ , so that  $x \in \xi$  a.s. by the closedness of  $\xi$  and so  $p_x = 1$ . Thus, the sum of  $p_{x_n}$  for each sequence  $\{x_n\}$  in a neighbourhood of x is finite. This rules out the existence of uncountably many y with  $p_y > 0$  in any neighbourhood of x. Indeed, then  $\{y : p_y \ge 1/n\}$  is finite for all n, and so the union of such sets is countable.  $\Box$ 

It is known that  $\mathcal{F}$  is compact and completely regular in the Fell topology that generates the Effros  $\sigma$ -algebra; see [14], Theorem 17.V.3 and [22]. However, Corollary 2.7 is not applicable, since functions  $\mathbb{1}_{x,y\in F}$ ,  $F \in \mathcal{F}$ , generating the vector space G, do not generate the Effros  $\sigma$ -algebra on  $\mathcal{F}$ .

It is known ([22], Theorem 1.2.6) that the  $\sigma$ -algebra generated by G on the family of *regular closed* sets (that coincide with closures of their interiors) coincides with the trace of the Effros  $\sigma$ -algebra on the family of regular closed sets. However, the family of regular closed sets is no longer compact in the Fell topology. Furthermore, indicator functions are not continuous in the Fell topology, so it is again not possible to appeal to Corollary 2.7 or explicitly check the upper semicontinuity condition required in Daniell's theorem.

In order to describe a useful family G of functionals acting on sets, consider a  $\sigma$ -finite measure  $\nu$  on X and define

(4.2) 
$$g_h(F) = \int_{F \times F} h(x, y) \nu(dx) \nu(dy)$$

for all measurable  $F \subset \mathbb{X}$  and h from the family  $\mathscr{C}_0$  of symmetric continuous functions with compact support in  $\mathbb{X} \times \mathbb{X}$ . A function  $p_{x,y}$ ,  $x, y \in \mathbb{X}$ , generates a functional acting on  $g_h$  as

(4.3) 
$$\Phi(\mathbf{g}_h) = \int_{\mathbb{X}\times\mathbb{X}} p_{x,y} h(x, y) \nu(dx) \nu(dy).$$

The function  $p_{x,y}$  is said to be *weakly realisable* as the two-point covering function if there exists a random set  $\xi$  (or the corresponding random indicator function) such that  $\xi$  is almost surely measurable and  $\mathbf{Eg}_h(\xi) = \Phi(\mathbf{g}_h)$  for all  $h \in \mathscr{C}_0$ . By approximating the atomic masses at two points with continuous functions, it is easy to see that the weak realisability is equivalent to  $\Phi(\mathbf{g}_{x,y}) = p_{x,y}$  for  $v \otimes v$ -almost all (x, y), in contrast to the *strong realisability* requiring this equality everywhere. The strong and weak realisability do not coincide in general. For instance, a nonpositive function which vanishes almost everywhere, but takes some negative values is weakly realisable by the empty set, but not strongly realisable. Nevertheless, in the case of a *stationary* random regular closed set  $\xi$  in  $\mathbb{R}^d$  and the Lebesgue measure v, the strong and weak realisability properties coincide; see Theorem 4.7.

In view of the required continuity property of functions from G, it is essential to ensure that  $g_h(F)$ ,  $F \in \mathcal{F}$ , defined in (4.2) is continuous in the Fell topology. Note that it is not the case for most nontrivial measures  $\nu$ , even for the Lebesgue measure. The continuity holds only on some subfamilies of  $\mathcal{F}$  considered in the following sections.

4.3. Neighbourhoods of closed sets. For simplicity, in the following consider random sets in the Euclidean space, that is, assume that  $\mathbb{X} = \mathbb{R}^d$  with Euclidean metric **d**.

Let  $\mathcal{F}^{\varepsilon}$  be the family of  $\varepsilon$ -neighbourhoods of closed sets in  $\mathbb{R}^d$ , that is,  $\mathcal{F}^{\varepsilon}$  consists of  $F^{\varepsilon} = \{x : \mathbf{d}(x, F) \le \varepsilon\}$  for all  $F \in \mathcal{F}$  and also contains the empty set. The vector space G is generated by constants and the functions  $g_h$  defined by (4.2) with the Lebesgue measure  $\nu$ .

LEMMA 4.5. The space  $\mathcal{F}^{\varepsilon}$  with the Fell topology is compact and, for each  $h \in \mathcal{C}_{o}$ , the function  $g_{h}$  is continuous on  $\mathcal{F}^{\varepsilon}$ .

PROOF. Recall that the upper limit of a sequence of sets  $\{F_n\}$  is the set of all limits for sequences  $\{x_{n_k}\}$  such that  $x_{n_k} \in F_{n_k}$  for all k, while the lower limit is the set of all limits for convergent sequences  $\{x_n\}$  such that  $x_n \in F_n$  for all n. The sequence of closed sets converges in the Fell topology if its upper and lower limits coincide.

If  $F_n = F_{n,0}^{\varepsilon} \in \mathcal{F}^{\varepsilon}$  converges to F in the Fell topology, then we can assume without loss of generality (by passing to subsequences) that  $F_{n,0}$  converges to  $F_0$ , so that  $F = F_0^{\varepsilon}$  and  $F \in \mathcal{F}^{\varepsilon}$ . Thus,  $\mathcal{F}^{\varepsilon}$  is a closed subset of  $\mathcal{F}$  and so is compact, since  $\mathcal{F}$  is compact itself.

Consider a nonnegative  $h \in \mathscr{C}_0$  supported by a ball  $B_R$  centred at the origin with sufficiently large radius R. If  $F_n \to F$  in the Fell topology, then the upper limit of  $(F_n \cap B_R)$  is a subset of  $(F \cap B_R)$ . Thus,  $g_h(F) = g_h(F_n \cap B_R) \le g_h((F \cap B_R)^{\delta})$ for any  $\delta > 0$  and sufficiently large n, so that  $g_h$  is upper semi-continuous on  $\mathcal{F}$ and so on  $\mathcal{F}^{\varepsilon}$ .

In order to prove the lower semi-continuity of  $g_h$  on  $\mathcal{F}^{\varepsilon}$  assume  $F_n = F_{n,0}^{\varepsilon} \rightarrow F = F_0^{\varepsilon}$  and  $F_{n,0} \rightarrow F_0$ . Fix  $\delta > 0$ . Then the lower limit of  $F_{n,0} \cap B_{r+\delta}$  includes  $F_0 \cap B_R$ . Indeed, if  $x \in F_0 \cap B_R$ , then  $x_n \rightarrow x$  for  $x_n \in F_{n,0}$  and so  $x_n \in B_{R+\delta}$  for all sufficiently large *n*. Thus, for sufficiently large *n*, we have

$$(F_0 \cap B_R) \subset (F_{n,0} \cap B_{R+\delta})^{\delta}.$$

Taking  $(\varepsilon - \delta)$ -neighbourhoods of the both sides yields that

$$(F_0 \cap B_R)^{\varepsilon - \delta} \subset (F_{n,0} \cap B_{R+\delta})^{\varepsilon} \subset (F_n \cap B_{R+\delta+\varepsilon}).$$

If  $x \in F_0^{\varepsilon-\delta} \cap B_{R-\varepsilon+\delta}$ , then there is a point  $y \in F_0$  with  $\mathbf{d}(x, y) \le \varepsilon - \delta$ , in particular  $y \in B_R$ . Thus,

$$(F_0^{\varepsilon-\delta}\cap B_{R-\varepsilon+\delta})\subset (F_{n,0}\cap B_R)^{\varepsilon-\delta}.$$

Taking *r* sufficiently large yields that  $g_h(F_n) \ge g_h(F_0^{\varepsilon-\delta})$ . Since the interior of *F* equals  $\bigcup_{\delta>0} F_0^{\varepsilon-\delta}$ , the Lebesgue theorem yields that  $g_h(F_0^{\varepsilon-\delta}) \to g_h(F)$  as  $\delta \to 0$ , that is,  $g_h$  is lower semi-continuous on  $\mathcal{F}^{\varepsilon}$ .

For a nonpositive function h with compact support, the result follows by applying the above argument to its positive and negative parts.  $\Box$ 

THEOREM 4.6. A function  $p_{x,y}$ ,  $x, y \in \mathbb{R}^d$ , is weakly realisable by a random closed set  $\xi$  with realisations in  $\mathcal{F}^{\varepsilon}$  for given  $\varepsilon > 0$  if and only if

$$\Phi(\mathbf{g}_h) \ge \inf_{F \in \mathcal{F}^{\varepsilon}} \mathbf{g}_h(F), \qquad h \in \mathscr{C}_0,$$

where  $\Phi(g_h)$  is given by (4.3).

PROOF. In view of the continuity of  $g_h$  established in Lemma 4.5, it suffices to refer to Corollary 2.7.  $\Box$ 

In order to handle random sets with realisations from the space  $\mathcal{F}^0 = \bigcup_{\varepsilon>0} \mathcal{F}^{\varepsilon}$ , we need the regularity modulus  $\chi(F)$  defined as the infimum of  $\varepsilon > 0$  such that  $F \in \mathcal{F}^{1/\varepsilon}$  and  $\chi(F) = \infty$  if  $F \notin \mathcal{F}^0$ .

THEOREM 4.7. For any given r > 0, a function  $p_{x,y}$ ,  $x, y \in \mathbb{R}^d$ , is weakly realisable by a random closed set  $\xi$  such that  $\mathbf{E}\chi(\xi) \leq r$  if and only if

(4.4) 
$$\inf_{F \in \mathcal{F}^0} [\chi(F) - \mathsf{g}_h(F)] + \Phi(\mathsf{g}_h) \le r, \qquad h \in \mathscr{C}_0,$$

where  $\Phi(g_h)$  is given by (4.3). If, additionally,  $p_{x,y}$  is an even continuous function of x - y, then  $p_{x,y}$  is strongly realisable by a stationary random closed set  $\xi$ .

**PROOF.** Function  $\chi$  is lower semi-continuous, since  $\{F \in \mathcal{F} : \chi(F) \le c\} = \mathcal{F}^{1/c}$  is closed for all c > 0. Furthermore,

(4.5) 
$$\left\{F \in \mathcal{F} : \chi(F) \le g_h(F)\right\} \subset \left\{F \in \mathcal{F} : \chi(F) \le c\right\},$$

where  $c = \int |h(x, y)| v(dx) v(dy)$  is a finite upper bound for  $g_h(F)$ . The left-hand side of (4.5) is compact, since  $g_h$  is continuous on  $\mathcal{F}^{1/c}$  by Lemma 4.5 and the right-hand side of (4.5) is compact. Thus,  $\chi$  is a regularity modulus and the result follows from Theorem 2.6 and (2.2).

Note that the regularity modulus  $\chi$  is invariant for the group  $\Theta$  of translations of  $\mathbb{R}^d$ . By Theorem 2.10(ii),  $\xi$  can be chosen to be stationary. In order to confirm the strong realisability, it remains to show that the covariance function of a stationary regular closed random set is continuous.

Since  $\chi(\xi)$  is integrable,  $\xi \in \mathcal{F}^0$ , so that  $\xi$  is almost surely regular closed and its boundary  $\partial \xi$  has a.s. vanishing Lebesgue measure. By Fubini's theorem, almost every point *x* belongs to the boundary of  $\xi$  with probability zero, and so  $\mathbf{P}\{x \in \partial \xi\} = 0$  for all *x* in view of the stationarity property.

Let  $\mathbf{P}\{x, y \in \xi\}$  be the covariance function of  $\xi$ . Take  $x, y \in \mathbb{R}^d$ , and  $(x_n, y_n)$  that converges to (x, y). Since with probability 1, x does not belong to  $\partial \xi$ ,  $\mathbb{1}_{x \in \xi}$  is almost surely equal to  $\mathbb{1}_{x \in \text{Int}(\xi)}$  for the interior  $\text{Int}(\xi)$  of  $\xi$  and so  $\mathbb{1}_{x_n \in \xi}$  almost surely converges to  $\mathbb{1}_{x \in \xi}$ . Similarly,  $\mathbb{1}_{y_n \in \xi} \to \mathbb{1}_{y \in \xi}$  a.s., whence the product converges too  $\mathbb{1}_{x_n \in \xi, y_n \in \xi} \to \mathbb{1}_{x \in \xi, y \in \xi}$ . The Lebesgue theorem yields that  $\mathbf{P}\{x_n, y_n \in \xi\} \to \mathbf{P}\{x, y \in \xi\}$ . Since  $p_{x,y}$  and  $\mathbf{P}\{x, y \in \xi\}$  are both continuous and coincide almost surely, they are equal everywhere. The continuity of  $\mathbf{P}\{x, y \in \xi\}$  can be also obtained by referring to a result of [23] saying that the capacity functional of each stationary regular closed random set is continuous in the Hausdorff metric.  $\Box$ 

4.4. Convexity restrictions. The family C of convex closed sets in  $\mathbb{R}^d$  (including the empty set) is closed in the Fell topology and it is easy to see that the function  $g_h$  given by (4.2) is continuous on C. Corollary 2.7 yields that  $p_{x,y}$  is weakly realisable for a convex random closed set if and only if

$$\Phi(\mathsf{g}_h) \ge \inf_{F \in \mathcal{C}} \mathsf{g}_h(F)$$

for the functional  $\Phi(g_h)$  given by (4.3).

Let  $\mathcal{P}$  be the *convex ring* in  $\mathbb{R}^d$ , that is, the family of finite unions of compact convex subsets of  $\mathbb{R}^d$ . For  $F \in \mathcal{P}$ , let  $\chi(F)$  be the smallest number k, such that F can be represented as the union of k convex compact sets.

THEOREM 4.8. Let  $\Phi$  be linear functional defined by (4.3). Fix any r > 0. Then there is a random closed set  $\xi$  with realisations in  $\mathcal{P}$  such that  $\mathbf{Eg}_h(\xi) = \Phi(\mathbf{g}_h)$  for all  $h \in \mathcal{C}_0$  and  $\mathbf{E}\chi(\xi) \leq r$  if and only if

$$\inf_{F \in \mathcal{P}} [\chi(F) - g_h(F)] + \Phi(g_h) \le r, \qquad h \in \mathscr{C}_0.$$

PROOF. The family  $\mathcal{P}_k$  of unions of at most k convex compact sets is closed in  $\mathcal{F}$  and so is compact, whence  $\chi$  is lower semi-continuous. It is easily seen that  $g_h$  is continuous on convex compact sets, and so is also continuous on  $\mathcal{P}_k$ . Thus,  $g_h$  is  $\chi$ -regular and Theorem 2.6 applies.  $\Box$ 

If  $\mathbb{X} = [0, 1]$ , then  $\mathcal{P}$  is be the family of finite unions of segments in [0, 1]. The number of convex components of  $F \subset [0, 1]$  is the variation of its indicator function,

$$\chi(F) = \sup \sum_{i=0}^{n-1} |\mathbb{1}_{t_i \in F} - \mathbb{1}_{t_{i+1} \in F}|,$$

where the supremum is taken over partitions  $0 = t_0 \le t_1 \le \cdots \le t_n = 1$ . The quantity

$$\mathsf{v}(F) = \sup_{\varphi \in \mathscr{C}^1, 0 \le \varphi \le 1} \int_F \varphi'(x) \, dx,$$

where  $\mathscr{C}^1$  is the family of differentiable functions on [0, 1], captures the number of components of *F* with nonempty interiors, in particular  $v(F) \le \chi(F)$ . Remark that v is not a regularity modulus, because a set *F* with small v(F) can contain an arbitrarily large number of isolated singletons.

THEOREM 4.9. If  $p_{x,y}$  is a function of  $x, y \in [0, 1]$  such that

(4.6) 
$$\sup_{\varphi \in \mathscr{C}^1, 0 \le \varphi \le 1} \int_{\mathbb{X} \times \mathbb{X}} p_{x,y} \varphi'(x) \varphi'(y) \, dx \, dy = \infty,$$

then there is no random closed set  $\xi$  satisfying  $\mathbf{E}\chi(\xi)^2 < \infty$  having  $p_{x,y}$  as its two-point covering function.

**PROOF.** Let H be the family of functions  $h(x, y) = \varphi'(x)\varphi'(y)$  for  $\varphi \in \mathscr{C}^1$  with  $0 \le \varphi \le 1$ . Then

$$\mathsf{v}(F)^2 = \sup_{h \in \mathsf{H}} \int_{\mathbb{X} \times \mathbb{X}} \mathbb{1}_{x, y \in F} h(x, y) \, dx \, dy = \sup_{h \in \mathsf{H}} \mathsf{g}_h(F).$$

Theorem 2.6 implies that  $\Phi$  is realisable by a random closed set  $\xi$  with  $\mathbf{E}\chi(\xi)^2 < \infty$  if and only if

$$\sup_{h\in\mathscr{C}_0} \left[\inf_{F\in\mathscr{X}} \left[\chi(F)^2 - \mathsf{g}_h(F)\right] + \Phi(\mathsf{g}_h)\right] < \infty.$$

It implies in particular

$$\sup_{h\in\mathsf{H}}\left[\inf_{F\in\mathcal{X}}\left[\chi(F)^2-\mathsf{g}_h(F)\right]+\Phi(\mathsf{g}_h)\right]<\infty.$$

Since  $\chi(F)^2 \ge v(F)^2 \ge g_h(F)$  for  $h \in H$ , this condition would imply that

$$\sup_{h\in\mathsf{H}}\Phi(\mathsf{g}_h)<\infty,$$

contradicting (4.6). Thus  $\Phi$  is not realisable.  $\Box$ 

Further results on realisability of random sets can be found in [6], where it is shown that by relaxing the closedness assumption it is possible to split the positivity and regularity conditions as it was the case in Section 3.3.

5. Contact distribution functions for random sets. Results from Section 4 concern realisability of the two-point covering probabilities, which are closely related to the values of the capacity functional (hitting probabilities) on two-point sets. Here, we consider the realisability problem for a capacity functional defined on the family of balls in  $\mathbb{R}^d$ . If *T* is the capacity functional of a random closed set  $\xi$ , then

$$T(B_R(x)) = \mathbf{P}\{\xi \cap B_R(x) \neq \emptyset\}$$

is closely related to the spherical contact distribution function  $\mathbf{P}\{\mathbf{d}(x,\xi) \le R | x \notin \xi\}$ ,  $R \ge 0$ , which is the cumulative distribution function of the distance between  $\xi$  and x given that  $x \notin \xi$ .

THEOREM 5.1. A function  $\tau_x(R)$ ,  $R \ge 0$ ,  $x \in A \subset \mathbb{R}^d$ , is realisable as  $T(B_R(x))$  for a random closed set  $\xi$  if and only if

(5.1) 
$$\Phi(\mathbf{g}) = \sum_{i=1}^{m} a_i \tau_{x_i}(R_i) \ge 0$$

for all  $m \ge 1, x_1, \ldots, x_m \in A$  and  $R_1, \ldots, R_m \ge 0$ , such that the function

(5.2) 
$$g(F) = \sum_{i=1}^{m} a_i \mathbb{1}_{B_{R_i}(x_i) \cap F \neq \emptyset} \ge 0, \qquad F \in \mathcal{F}$$

is nonnegative.

PROOF. The necessity is evident.

Sufficiency. Let G be the vector space generated by constants and functions  $g_{h,x}(F) = h(\mathbf{d}(x, F)), F \in \mathcal{F}$ , where  $\mathbf{d}(x, F)$  is the distance from  $x \in \mathbb{R}^d$  to the nearest point of *F*, and *h* is a continuous function on  $\mathbb{R}$  with bounded support. The functions  $g_{h,x}$  are all continuous in the Fell topology, since the Fell topology in  $\mathbb{R}^d$  coincides with the topology of pointwise convergence of distance functions  $\mathbf{d}(x, F)$  for  $x \in \mathbb{R}^d$ ; see [22], Theorem B.12.

It suffices to show that  $\Phi$  is positive on G. Let  $g(F) = \sum_{i=1}^{m} a_i h_i(\mathbf{d}(x_i, F))$ . Uniform approximation of  $h_1, \ldots, h_m$  by step functions on their supports yields a function  $\hat{g}$  of the form (5.2) so that  $\hat{g}(F) \ge -\varepsilon$  for some  $\varepsilon > 0$ . Letting  $\varepsilon \downarrow 0$  and using (5.1) yield that

$$\Phi(\mathbf{g}) = \sum_{i=1}^{m} a_i \int h_i(t) \, d\tau_{x_i}(t) \ge 0.$$

If  $\tau_x(R) = \tau(R)$  does not depend on *x*, it may be possible to realise it as the contact distribution function of a stationary random closed set. If the argument *x* of  $\tau_x(R)$  takes only a single value, then the necessary and sufficient condition on  $\tau_x(\cdot)$  is that it is a nondecreasing right-continuous function with values in [0, 1], that is, the cumulative distribution function of a sub-probability measure on  $\mathbb{R}_+$ . The following result concerns the case of *x* taking two possible values.

THEOREM 5.2. Let  $x_1, x_2 \in \mathbb{R}^d$ , with  $l = ||x_1 - x_2||$ , and let  $\tau_{x_1}$  and  $\tau_{x_2}$  be cumulative distribution functions of two sub-probability measures on  $\mathbb{R}_+$ . Then there exists a random closed set  $\xi$  such that  $\tau_{x_i}(R) = T(B_R(x_i))$  for  $r \ge 0$  and i = 1, 2 if and only if for all  $r \ge 0$ 

(5.3) 
$$\tau_{x_1}(\max(R-l,0)) \le \tau_{x_2}(R) \le \tau_{x_1}(R+l).$$

PROOF. Necessity. Let  $\xi$  be a random closed set with  $\tau_{x_i}(R) = T(B_R(x_i))$ . Let  $a_1$  and  $a_2$  be random points such that  $a_1, a_2 \in \xi$  a.s. and  $R_i = \mathbf{d}(x_i, a_i) = \mathbf{d}(x_i, \xi)$ , i = 1, 2, have cumulative distribution functions  $\tau_{x_1}$  and  $\tau_{x_2}$ , respectively. Then  $|R_1 - R_2| \le l$ . Indeed, if, for instance,  $R_1 > R_2 + l$ , then  $a_2$  is nearer to  $x_1$  than  $a_1$  contrary to the assumption. Thus  $R_1 \le R$  implies  $R_2 \le R + l$ , so that  $\tau_{x_1}(R) \le \tau_{x_2}(R+l)$ . The symmetry argument with  $x_1$  and  $x_2$  interchanged yields (5.3).

Sufficiency. Define two random variables  $R_1$  and  $R_2$  as inverse functions to  $\tau_{x_1}$  and  $\tau_{x_2}$  applied to a single uniform random variable, so that (5.3) yields that  $|R_1 - R_2| \le l$  a.s. This means that none of the balls  $B_{R_1}(x_1)$  and  $B_{R_2}(x_2)$  lies in the interior of the other one. Now construct random closed set  $\xi$  consisting of two points:  $a_1$  on the boundary of  $B_{R_1}(x_1)$  but outside of the interior of  $B_{R_2}(x_2)$  and  $a_2$  on the boundary of  $B_{R_2}(x_2)$  but outside of the interior of  $B_{R_2}(x_1)$ . Then  $a_1$  is nearest to  $x_1$  and  $a_2$  is nearest to  $x_2$  with given distributions of the distance.  $\Box$ 

### APPENDIX: A COMBINATORIAL LEMMA

Recall that  $P_t(\mathbb{X})$  denotes the *packing number* of  $\mathbb{X}$  with metric **d**, that is, the maximum number of points in the space  $\mathbb{X}$  with pairwise distance exceeding *t*; see [20], page 78.

LEMMA A.1. If  $Y = \sum \delta_{x_i}$  is a counting measure of total mass n, then for all t > 0,

$$\sum_{i\neq j} \mathbb{1}_{\mathbf{d}(x_i, x_j) \le t} \ge n \bigg( \frac{n}{P_t(\mathbb{X})} - 1 \bigg).$$

PROOF. Denote

$$n(Y, x_i) = Y(B_t(x_i)) - 1,$$

where  $B_t(x_i)$  is the closed ball of radius t centred at  $x_i$ . Furthermore, denote

$$\mathbf{g}_{h_t}(Y) = \sum_{i \neq j} \mathbb{1}_{\mathbf{d}(x_i, x_j) \le t}$$

Then

$$g_{h_t}(Y - \delta_{x_i}) = g_{h_t}(Y) - 2n(Y, x_i),$$
  

$$g_{h_t}(Y + \delta_{x_i}) = g_{h_t}(Y) + 2n(Y, x_i) + 2$$

Let  $x_i$  and  $x_j$  be two distinct points from the support of Y with  $\mathbf{d}(x_i, x_j) \le t$ . Assume that  $n(Y, x_i) < n(Y, x_j)$  or  $n(Y, x_i) = n(Y, x_j)$  with i < j and define

$$Y' = Y - \delta_{x_i} + \delta_{x_i}$$

obtained from *Y* by transferring a mass 1 from  $x_j$  to  $x_i$ . Call  $Y'' = Y - \delta_{x_j}$ . Remark that  $n(Y'', x_i) = n(Y, x_i) - 1$  because  $\mathbf{d}(x_i, x_j) \le t$ . Since  $n(Y, x_j) \ge n(Y, x_i)$ ,

$$g_{h_t}(Y') = g_{h_t}(Y'') + 2n(Y'', x_i) + 2$$
  
=  $g_{h_t}(Y) - 2n(Y, x_j) + 2n(Y'', x_i) + 2$   
=  $g_{h_t}(Y) - 2n(Y, x_j) + 2n(Y, x_i) - 2 + 2$   
 $\leq g_{h_t}(Y).$ 

Furthermore,  $n(Y', x_i) = n(Y, x_i)$  because the transferred mass remains in the ball with centre  $x_i$  and radius t, and  $n(Y', x_j) = n(Y, x_j)$  as well. Thus,  $n(Y', x_i) \le n(Y', x_j)$ . Repeat the mass transfer from  $x_j$  to  $x_i$  until the mass at  $x_j$  disappears. Call the resulting counting measure  $Y_1$ .

Apply the same construction to  $Y_1$  and repeat it until there are no more distinct points at distance at most t. This happens in a finite time because the cardinality of the support of Y strictly decreases at each step.

The obtained counting measure  $\hat{Y}$  is supported by a set of points  $\{y_1, \ldots, y_q\}$  with pairwise distances exceeding t. Thus,

$$\mathsf{g}_{h_t}(Y) \ge \mathsf{g}_{h_t}(\widehat{Y}) = \sum_{i=1}^q m_i(m_i - 1),$$

where  $m_i = \widehat{Y}(\{y_i\})$ . Under the restriction  $\sum_{i=1}^{q} m_i = n$ , the minimal value  $\sum_i m_i (m_i - 1)$  is reached for  $m_i = n/q$ , whence

$$\mathsf{g}_{h_t}(Y) \ge n \bigg( \frac{n}{q} - 1 \bigg).$$

It remains to note that  $q \leq P_t(\mathbb{X})$ .  $\Box$ 

It is also possible to define a counting measure by placing masses from the interval [n/q, n/q + 1] at the points forming the packing net of X. Thus, there exists a counting measure Y such that

$$g_{h_t}(Y) \le n\left(\frac{n}{P_t(\mathbb{X})}+1\right).$$

Acknowledgements. The authors are grateful to John Quintanilla and Zbigniew Lipecki for literature hints at early stages of this work and to Tobias Kuna for comments on the preprint version. The comments of the referees and the Editor greatly inspired the authors to improve the readability of the paper. Raphael Lachieze-Rey is grateful to the University of Bern for hospitality.

#### REFERENCES

- ALIPRANTIS, C. D. and BORDER, K. C. (2006). Infinite Dimensional Analysis, 3rd ed. Springer, Berlin. MR2378491
- [2] BOURBAKI, N. (1989). General Topology. Springer, Berlin. Chapters 5–10. Translated from the French, reprint of the 1966 edition. MR0979295
- [3] DALEY, D. J. and VERE-JONES, D. (1988). An Introduction to the Theory of Point Processes. Springer, New York. MR0950166
- [4] DUDLEY, R. M. (2002). *Real Analysis and Probability*. Cambridge Univ. Press, Cambridge. MR1932358
- [5] FRISTEDT, B. and GRAY, L. (1997). A Modern Approach to Probability Theory. Birkhäuser, Boston, MA. MR1422917
- [6] GALERNE, B. and LACHIÈZE-REY, R. (2013). Random measurable sets and realisability problems. Unpublished manuscript.
- [7] HOLLEY, R. A. and STROOCK, D. W. (1978). Nearest neighbor birth and death processes on the real line. Acta Math. 140 103–154. MR0488380
- [8] JIAO, Y., STILLINGER, F. H. and TORQUATO, S. (2007). Modeling heterogeneous materials via two-point correlation functions: Basic principles. *Phys. Rev. E* (3) 76 031110. MR2365559
- KELLERER, H. G. (1964). Verteilungsfunktionen mit gegebenen Marginalverteilungen. Z. Wahrsch. Verw. Gebiete 3 247–270. MR0175158
- [10] KONDRATIEV, YU. G. and KUTOVIY, O. V. (2006). On the metrical properties of the configuration space. *Math. Nachr.* 279 774–783. MR2226411
- [11] KÖNIG, H. (1997). Measure and Integration: An Advanced Course in Basic Procedures and Applications. Springer, Berlin. MR1633615
- [12] KUNA, T., LEBOWITZ, J. L. and SPEER, E. R. (2007). Realizability of point processes. J. Stat. Phys. 129 417–439. MR2351408
- [13] KUNA, T., LEBOWITZ, J. L. and SPEER, E. R. (2011). Necessary and sufficient conditions for realizability of point processes. Ann. Appl. Probab. 21 1253–1281. MR2857448
- [14] KURATOWSKI, K. (1966). Topology, Vol. I. Academic Press, New York. MR0217751
- [15] LENARD, A. (1975). States of classical statistical mechanical systems of infinitely many particles. I. Arch. Ration. Mech. Anal. 59 219–239. MR0391830
- [16] LENARD, A. (1975). States of classical statistical mechanical systems of infinitely many particles. II. Characterization of correlation measures. *Arch. Ration. Mech. Anal.* 59 241–256. MR0391831

- [17] MARKOV, K. Z. (1995). On the "triangular" inequality in the theory of two-phase random media. Technical Report 89/1995, 159-166, Annuaire de'Universite de Sofia "St. Klimen Ohridski," Faculte de mathematiques et Informatique, Livre I—Mathematiques et Mécanique.
- [18] MATHERON, G. (1975). Random Sets and Integral Geometry. Wiley, New York. MR0385969
- [19] MATHERON, G. (1993). Une conjecture sur covariance d'un ensemble aleatoire. Cahiers de Géostatistique 107 107–113.
- [20] MATTILA, P. (1995). Geometry of Sets and Measures in Euclidean Spaces. Cambridge Univ. Press, Cambridge. MR1333890
- [21] MCMILLAN, B. (1955). History of a problem. J. Soc. Indust. Appl. Math. 3 119–128. MR0074724
- [22] MOLCHANOV, I. (2005). Theory of Random Sets. Springer, London. MR2132405
- [23] MOLCHANOV, I. S. (1989). On convergence of empirical accompanying functionals of stationary random sets. *Theory Probab. Math. Statist.* 38 107–109.
- [24] QUINTANILLA, J. A. (2008). Necessary and sufficient conditions for the two-point phase probability function of two-phase random media. Proc. R. Soc. Lond. Ser. A Math. Phys. Eng. Sci. 464 1761–1779. MR2403127
- [25] SCHNEIDER, R. and WEIL, W. (2008). Stochastic and Integral Geometry. Springer, Berlin. MR2455326
- [26] SHARAKHMETOV, SH. and IBRAGIMOV, R. (2002). A characterization of joint distribution of two-valued random variables and its applications. J. Multivariate Anal. 83 389–408. MR1945960
- [27] SHEPP, L. A. (1963). On positive-definite functions associated with certain stochastic processes. Technical Report 63-1213-11, Bell Telephone Laboratories, Murray Hill, NJ.
- [28] SHEPP, L. A. (1967). Covariances of unit processes. In Proc. Working Conf. Stochastic Processes 205–218. Santa Barbara, California, CA.
- [29] SILVERMAN, R. J. (1956). Invariant linear functions. Trans. Amer. Math. Soc. 81 411–424. MR0082065
- [30] STOYAN, D., KENDALL, W. S. and MECKE, J. (1995). Stochastic Geometry and Its Applications, 2nd ed. Wiley, Chichester.
- [31] TORQUATO, S. (1999). Exact conditions on physically realizable correlation functions of random media. J. Chem. Phys. 111 8832–8837.
- [32] TORQUATO, S. (2002). Random Heterogeneous Materials. Springer, New York. MR1862782
- [33] TORQUATO, S. (2006). Necessary conditions on realizable two-point correlation functions of random media. *Indus. Eng. Chem. Res.* 45 6923–6928.
- [34] TORQUATO, S. and STELL, G. (1982). Microstructure of two-phase random media. I. The n-point probability functions. J. Chem. Phys. 77 2071–2077. MR0668092
- [35] VULIKH, B. Z. (1967). Introduction to the Theory of Partially Ordered Spaces. Wolters-Noordhoff, Groningen. MR0224522
- [36] WHITTLE, P. (1992). Probability via Expectation. Springer, New York. MR1163374

LABORATOIRE MAP 5 UNIVERSITÉ PARIS DESCARTES 45 RUE DES SAINTS-PÈRES 75006 PARIS FRANCE E-MAIL: lr.raphael@gmail.com INSTITUTE OF MATHEMATICAL STATISTICS AND ACTUARIAL SCIENCE UNIVERSITY OF BERN SIDLERSTRASSE 5 3012 BERN SWITZERLAND E-MAIL: ilya@stat.unibe.ch